A Comprehensive Review on Rectifiers, Linear Regulators, and Switched-Mode Power Processing Techniques for Biomedical Sensors and Implants Utilizing in-Body Energy Harvesting and External Power Delivery

Sourov Roy[®], *Graduate Student Member, IEEE*, A N M Wasekul Azad[®], *Graduate Student Member, IEEE*, Somen Baidya[®], *Student Member, IEEE*, and Faisal Khan, *Senior Member, IEEE*

Abstract—This article reviews the state-of-the-art power conditioning techniques to process harvested energy from the human body and any external environmental sources surrounding the human body for powering biomedical implants and sensors. The fundamental focus of this article is to highlight the necessity of power conditioning circuits to energize implanted biomedical circuits. In addition, the underlying challenges in power conversion modules used in these low-power circuits have been discussed in detail. Power conditioning techniques for biomedical implants heavily rely on the type of sources and implants, thereby we aim to provide an elaborate discussion on the operating principle of individual technique through this review. Besides, we have also explored the suitability of each individual power processing technique inside the human body with great details. Finally, a comprehensive discussion on the limitations of existing power conditioning techniques has been presented, and the scopes to improve or mitigate those limitations have been discussed as well. The goal of this work is to present a complete guideline for future researchers to downselect a compatible power conditioning technique for a specific biomedical application.

Index Terms—AC–DC, biomedical implants, cold-start, control, dc–dc, energy harvesting, linear regulator, maximum power extraction, micro-electromechanical system (MEMS), power converter, rectifiers, sensor, switched capacitor (SC), switched inductor, wireless power delivery.

I. INTRODUCTION

MPLANTABLE medical devices (IMDs) used for clinically assisting patients using a combination of physiological

Manuscript received October 20, 2020; revised January 25, 2021 and March 23, 2021; accepted April 7, 2021. Date of publication April 23, 2021; date of current version July 30, 2021. Recommended for publication by Associate Editor D. O. Neacsu. (*Corresponding author: Sourov Roy.*)

Sourov Roy is with the Missouri Institute for Defense and Energy, University of Missouri—Kansas City, Kansas City, MO, 64110, USA (e-mail: srdh9@mail.umkc.edu).

A N M Wasekul Azad, Somen Baidya, and Faisal Khan are with the Department of Computer Science and Electrical Engineering, University of Missouri– Kansas City, Kansas City, MO 64110 USA (e-mail: aax9c@mail.umkc.edu; sbg58@mail.umkc.edu; faisal.khan@umkc.edu).

Color versions of one or more figures in this article are available at https: //doi.org/10.1109/TPEL.2021.3075245.

Digital Object Identifier 10.1109/TPEL.2021.3075245

parameter sensing and varied methods of required stimulus incentives are becoming increasingly popular nowadays [1]. IMDs encompass a diversified class of devices in terms of size and driving power requirement, starting from bulky pacemaker to miniature neural stimulator, and from power-hungry total artificial heart (TAH)/left ventricular assist device (LVAD) to low power bone growth stimulator. The invasive installation nature (i.e., major surgery) and associated health risks of the IMDs often call for prolonged sustainability so that patients do not frequently need to undergo the same surgical process. Traditional energy storage devices (e.g., nonrechargeable battery) has limited energy density [2], and thus, they cannot support powerhungry circuits indefinitely. Therefore, alternative solutions are needed to meet the extended energy requirements of the IMDs to provide a prolonged lifetime. As an in-body application, the implant must not cause any discomfort and interference with the daily activities of the patient, which leads to the miniaturization of its footprint. In order to miniaturize implants and minimize the number of invasive battery-replacement surgeries during a patient's lifetime, autonomous operation of the IMDs facilitated by in-body energy harvesting and/or external power delivery is being widely perceived as an indispensable alternative to the traditional battery-powered operation of these devices. That being said, sporadic availability of some ambient energy harvesting sources on a daily basis requires an intermediate energy storage device incorporated in the IMDs. The potential energy storage devices appropriate for powering the IMDs vary with the intended applications, type of energy sources, and implant location inside a human body. The power requirement of IMDs can vary from as low as few microwatts (e.g., bone growth stimulators, etc.) to as high as \sim 30 W (e.g., TAH/LVAD, etc.). A location-based list of several IMDs along with their respective power requirements is illustrated in Table I [3]-[21].

Harvesting energy from the human body and the surrounding include kinetic, triboelectric, thermoelectric (TE), glucose biofuel, photovoltaic (PV), etc. Delivering energy to IMDs from outside the body includes inductive power transfer (IPT), ultrasonic energy transfer (UET), optical link, capacitive link,

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

IMDs	Bone growth stimulator	Cardiac pacemaker	Drug pump	Body area network	Cochlear implant	Retinal stimulator	Neural stimulator	MCSD-LVAD /TAH [*]
References	[3], [4]	[5]-[7]	[8], [9]	[10], [11]	[12]-[14]	[15]-[17]	[18], [19]	[20], [21]
Power required	5 μW ~20 μW	<100 µW	100 μW ~2 mW	100 μW~ 5 mW	600 μW ~40 mW	1 mW ~100 mW	10 mW ~200 mW	5-30 W
Power level	Ultra-low	Ultra-low	Ultra- low/low	Ultra- low/low	Low	Low	Low	High
Location in the body	Bones undergoing fractures, and fusions	Right ventricle/ right atrium	Mostly at desired body site	Desired body site	Tympanic duct of the inner ear	Outer/internal surface of the retina	Cerebellum/ epidural space	Left ventricle/ middle mediastinum

 TABLE I

 POWER REQUIREMENTS FOR IMDS ALONG WITH THEIR LOCATIONS

Note: *MCSD-mechanical circulatory support device; LVAD-left ventricular assist device; TAH-total artificial heart.

and other technologies yet to be explored. Despite having the advantage of autonomous and regenerative power supply, IMDs that are powered by ambient sources experience numerous challenges during implementation. These challenges arise from the extreme size constraint, limitation of the available energy resources, and unstable power/voltage levels generation from the environment. Moreover, power/voltage level requirements for the IMDs are application-specific. For example, PV cells and electromagnetic (EM) harvesters produce very low voltages (typically less than 1 V), while electrostatic (ES) devices can generate voltages at over 100 V [22]. Various sensor implants and digital complementary metal-oxide-semiconductor (CMOS) based auxiliary electronic circuits operate at or below 1.8 V, while the majority of the functional electrical stimulators and LVAD/TAH implants require higher compliance voltages ranging from 2.5 to 38 V [10], [18]. Therefore, voltage up as well as downconversion may be required depending upon the types of energy harvesters and implants powered by them. In addition, inductive and ultrasonic power delivery and the majority of the energy harvesting sources (e.g., kinetic, triboelectric, etc.) generate ac voltage in general, although a few energy harvesting sources, such as solar/PV, microbial fuel cell (MFC), TE, etc., produce dc voltage as well [1]. The former requires an ac-to-dc as well as dc-to-dc voltage conversion, while the latter only needs dc-to-dc voltage conversion so that the energy-harvesting source can comply with the implant load requirement. Therefore, a stand-alone power management interface is necessary to handle the unpredictable power/voltage levels extracted from the environment, rectify, and/or convert these voltages to proper voltage levels for powering the implantable circuits. The interface should also provide the functionality to extract the maximum possible power from the harvesting source. In addition, the power management circuit should have a low-voltage cold-start functionality when the harvester voltage levels fall below the circuit threshold values. Therefore, the basic power electronics topology for such a power management interface system between the energy harvester and the implant load electronics generally follows the architecture that is shown in Fig. 1 [22].

Any power conversion circuit used in biomedical applications is subjected to design tradeoffs. However, these specifications are somewhat different in higher power circuits than ultralow and low-power circuits. This topic has been discussed in Section II. Fig. 2 shows the classification of the power conditioning



Fig. 1. Power electronics topology for energy harvesting system of IMDs [22].



Fig. 2. Families of power conditioning techniques used in IMDs.

techniques used in IMDs. The power conversion circuit can be a simple rectifier, a single-stage direct ac-dc converter, or a dc-dc converter. The general category of a rectifier that performs ac-to-dc voltage conversion without any voltage conditioning varies from simple passive rectifiers to active rectifiers. A passive full-wave rectifier can be made from junction-based diodes (p-n/Schottky) or diode-connected MOSFETs, whereas an active rectifier uses comparator-controlled MOSFETs in place of diodes/diode-connected MOSFETs [23], [24]. A brief review of these rectifier circuits is provided in Section III. In addition, rectification as well as voltage conditioning using a singlestage direct ac-dc converter has been discussed in Section IV, which includes self-commutated capacitor-based converters, also known as voltage multipliers (VM), and ac-dc switch-mode converters containing switched inductors. Section V presents a detailed overview of the existing dc-dc voltage converters, such as low drop-out (LDO) regulators and switched-mode regulators involving one or more energy storage components.

LDO has been traditionally used for voltage down-conversion using a continuous-time circuit with a dissipative element [25]. Switched-mode regulators can be configured in both step-down (buck) and step-up (boost) configurations. The energy can be stored either in inductors, yielding to a switched-inductor converter, or in a capacitor, leading to a switched capacitor (SC) converter [26]. Section VI provides an insight into the suitability of different types of power converters based on target applications while highlighting the strengths and weaknesses of each technique. Section VII discusses the start-up issues experienced by the power converter circuits and how these issues are addressed using power electronics. Section VIII explains today's technology gap and future research directions to counteract the weaknesses of power conversion techniques highlighted in this review. Finally, Section IX draws the conclusion.

II. POWER ELECTRONIC REQUIREMENTS IN BIOMEDICAL APPLICATIONS

Depending on the source, an energy harvesting technique may generate insufficient or incompatible voltage and current levels to the implantable load demand. Hence, further processing is required, which may involve a combination of rectification, voltage conversion, smoothing, and/or regulation depending on the specific application. The aforementioned processing techniques can be different from what is employed in traditional power electronic applications. In general, output voltage harvested from ambient energy resources is typically comparable to the threshold voltage of a semiconductor switch, while the power level required for a typical biomedical application is comparable to the quiescent power consumption in traditional power electronics applications [27]. Therefore, one of the fundamental challenges of incorporating power electronics-based converters in biomedical implants is achieving high efficiency at a low power level while maintaining a small form factor. Keeping that in focus, various passive and active rectification techniques have been developed for IMDs requiring steady (dc) voltage and current. LDOs and switched power converters can provide further voltage regulation for the target implant applications.

Conduction loss associated with LDOs and multiple loss mechanisms (e.g., control loss, leakage loss, switching loss, conduction loss, etc.) associated with the switched power converters in IMDs reduces the efficiency of the converter. To that end, effective loss mitigation schemes should be in place to improve the efficiency of the converter, which in turn ensures reliable operation of low-power devices such as IMDs. At this low power level, leakage and control losses can significantly degrade the overall efficiency of the system. Control losses can be minimized by simplifying the control loops that allow for scaling power consumption with output power. Leakage current can be mitigated through on-chip power gating and voltage scaling [27]. Unlike pulsewidth modulation (PWM), pulse-frequency modulation (PFM) leads to better power efficiency at low power levels since it reduces the switching loss by changing the operating frequency as the power decreases [26]. However, the variable frequency in the PFM technique makes noise-filtering difficult. Moreover, PFM may lead to a greater voltage ripple at the output. In a nutshell, when switching

loss mitigation is of concern, it is advantageous to opt for the PFM method at light load conditions, and the PWM method has the edge at heavy load conditions. Therefore, an intelligent switching scheme combining both PWM and PFM techniques depending on the load conditions can offer significant switching loss mitigation capability in biomedical implant applications [26], [28]. A significant mismatch exists between the levels of continuous harvested power and peak instantaneous power consumption of most implantable loads. Therefore, it is essential to design and integrate intermediate energy storage in the power converter interface of the implantable device to store sufficient energy that can be supplied when needed. Moreover, a cold-start/start-up circuit is also required to kick-start the conversion processes of the converter itself if the voltage levels of the low energy ambient sources fall below circuit threshold values.

While designing a power processing circuit for an IMD, the designer must ensure patient safety and security. Human exposures to electric, magnetic, and EM fields are allowed within a frequency range of 0 Hz to 300 GHz given that the thresholds of certain associated parameters specified by the safety standards (i.e., IEEE C95.1-2019/Cor 2-2020) are not violated [29]. A few examples of such parameters are internal electric field strength, specific absorption rate, and so on. According to the IEC-60601-1 standard, the use of galvanic isolation in an implantable circuit governs its maximum voltage limitation [30]. Electromagnetic interference (EMI) must also be considered as part of the power converter design for IMDs since imaging equipment, patient monitors, etc., have detection and signaling components that are very sensitive to EMI.

III. AC-DC CONVERSION: RECTIFICATION

Most of the IMDs require stable regulated dc input voltage, which can be delivered by a voltage converter. Since both the wireless power transfer (WPT) and kinetic energy harvesting techniques generate time-varying voltage, rectification is also needed apart from the voltage conversion. Rectification can be achieved either by passive or active topology.

A. Passive Rectification

The simplest way to achieve rectification is to use four junction-based diodes (p–n or Schottky) or diode-connected MOSFETs in different full-bridge rectifier (FBR) topologies, which are referred to as passive rectification, as shown in Fig. 3(a)–(g). The schematic of a conventional junction diode-based rectifier is depicted in Fig. 3(a) [8], [31]. This type of rectifier suffers from a significant voltage drop (~ 0.2 V or higher) in forward-bias mode, which contributes to $\sim 30\%$ power loss of an ac signal with an amplitude of 1 V [24]. This implies that the junction diode-based rectifier is not a feasible choice for low-voltage low-power biomedical applications. Moreover, in the context of integrated circuits, these junction diode-based rectifiers are not available in standard CMOS processes since the required fabrication process to integrate them in the submicrometer level CMOS chips is expensive [32].

To increase the efficiency at ultralow and low-power applications, the junction diodes of conventional rectifiers are usually



Fig. 3. Conventional ac-dc rectifier topologies. (a) Junction diode-based passive rectifier [8], [31]. (b) Diode-connected MOS passive rectifier [35]. (c) Gate-cross coupled passive rectifier [36]. (d) Differential fully gate-cross coupled passive rectifier [37]–[39]. (e) EVC passive rectifier [41]. (f) SVC passive rectifier [43]. (g) Active rectifier with cross-coupled PMOS switches [62]. (h) Synchronous rectifier using self-driven eGaN FETs [21].

replaced by MOSFETs in a diode-tied configuration using standard CMOS fabrication process [33]-[35]. These CMOS-based bridge rectifiers gained popularity in implantable biomedical applications due to their compatibility with the low-cost CMOS fabrication process, better efficiency at low voltage input signals, and so on. A simple MOSFET-based passive full-wave rectifier where the junction diodes are replaced with diode-connected MOS transistors is shown in Fig. 3(b). Efficiency at low input voltage due to the MOSFET threshold voltage $(V_{\rm th})$ drop, and therefore, the ideal maximum output dc voltage, is limited to $2 \times (V_{\rm in} - V_{\rm th}), V_{\rm in}$ being the maximum input signal amplitude [32]. That being said, the efficiencies of these MOSFET-based passive rectifiers are sensitive to $V_{\rm th}$. Moreover, this configuration may result in each MOSFET not being fully turned ON or OFF, which can increase the leakage current and voltage drop across the devices significantly [23].

To improve power conversion efficiency (PCE) in an integrated MOSFET-based FBR operation, two forms of gate crosscoupled rectifier technologies, namely gate cross-coupled rectifier and differential fully gate cross-coupled rectifier, are considered as the most popular choices [36]–[39]. In each signal cycle of the gate cross-coupled configurations, $V_{\rm th}$ of one or both of the main pass MOSFETs is replaced with the effective voltage drop across their complementary transistors, as shown in Fig. 3(c) and (d). However, this gate cross-coupled technology operates efficiently within a limited range of input power since it conducts in a reverse direction once the instantaneous value of the input ac signal becomes lower than the output dc voltage in every cycle. This periodic reverse leakage is exacerbated, and therefore, the PCE increases as the input power level grow. To improve the PCE at a higher power level, Ouda *et al.* [40] proposed a modified version of a gate cross-coupled rectifier with a self-biasing mechanism, which controls the conduction of the rectifying MOSFETs by raising their effective turn-ON voltage at high input power levels. This rectifier circuit was able to achieve more than 40% PCE and an input power range extension of more than 50% relative to the conventional gate cross-coupled rectifier.

The effect of $V_{\rm th}$ can also be mitigated by generating the required gate-bias voltage with the help of several $V_{\rm th}$ cancelation techniques, such as external- $V_{\rm th}$ -cancelation (EVC), internal- $V_{\rm th}$ -cancelation (IVC), and self- $V_{\rm th}$ -cancelation (SVC). To generate the required gate-bias voltage, the EVC scheme uses external battery/power sources [41], as shown in Fig. 3(e). In contrast, the IVC scheme utilizes an additional circuit powered by the rectifier itself [10], [42]. Both EVC and IVC schemes suffer from leakage current when the diode-connected MOSFET is reverse-biased. They also require additional hardware leading to additional cost and increased power consumption, which may become unsuitable for low power applications. Fig. 3(f) shows a schematic overview of the SVC rectifier, which can achieve better PCE at low input power conditions compared to EVC and IVC rectifiers [43]. SVC scheme applies the gate bias voltage generated from the output voltage of the rectifier itself, thus reducing the effective $V_{\rm th}$ of the MOS transistors. One potential drawback of using such an SVC scheme may arise when the effective $V_{\rm th}$ of the MOS transistor becomes very small or even negative due to the excessive dc bias voltage, resulting in a large leakage current and, hence, reducing the overall PCE.

In recent years, more focus is being driven into the body-effect compensation technique to mitigate the threshold voltage effect. To counteract the body-effect of the MOSFETs, several recent studies proposed to change their body-source potential using additional circuits [44], auxiliary transistors, and/or separated well transistors [6], [45], [46]. This body-effect elimination method also reduces the substrate leakage current, which is beneficial to the enhancement of the overall PCE of the rectifier circuit, and by incorporating this method, 80% PCE was reported very recently in [46].

Reducing/eliminating the impact of $V_{\rm th}$ is an active area of study in present times, and many new methods are being proposed to improve existing MOSFET-based passive rectification techniques, such as by integrating additional precharged floating gate transistors [47], [48] and bootstrapping circuitry [49], [50]. However, these techniques do not perform satisfactorily at a very low input voltage level and high operating frequency. As a result, more focus is being driven to hybrid type schemes combining two or more $V_{\rm th}$ cancelation schemes in order to increase efficiency at very low input voltage levels and high operating frequencies [42], [51], [52]; 70%–80% PCE is possible even for low input voltages, as shown by Hashemi *et al.* [51].

B. Active and Synchronous Rectification

Active rectification offers another alternative to further increase the efficiency of a rectifier by reducing the conduction losses of the pass transistors operating as switches in the deep triode region [23], [24]. Fig. 3(g) shows a simplified schematic of a typical active rectifier with cross-coupled PMOS switches. In active rectifier topology, the MOSFET is driven by a comparator and associated control circuitry (e.g., pulse generator, level shifter, etc.). The MOSFET together with its comparator is referred to as "active diode." The comparator continuously monitors the voltage drop across the MOSFET based on the source characteristics and load requirements and turns it ON or OFF in such a way that it can ensure minimal conduction loss and reverse leakage current. However, the comparator and control circuitry of active topologies presents additional power consumption. Fortunately, the overall power consumption is very low (<1 μ W) and independent of the MOSFET drain current, resulting in excellent PCE at high output power [49].

Several active rectifier topologies are reported in [32], [35], and [53]–[65] and the most basic topology of them includes two active diodes and two gate cross-coupled MOSFETs together with control circuitry and a self-body bias circuit(s). In reality, the delay time of the comparator and the gate-drive buffer limits the highest operating frequency and increases the reverse current of the rectifier, thereby compromising the output signal fidelity as well as the rectifier PCE and voltage conversion efficiency (VCE), especially at low input voltage (<1.5 V). To achieve high PCE and VCE by eliminating the reverse current, the power switches should be switched OFF right before the input ac signal goes below the output dc voltage. Comparators with constant/fixed artificial offset [54], [65], dynamic artificial offset [62], switched-offset biasing [57], etc., are used to compensate for the delay and to turn the power switches ON and OFF properly. A similar study by Rozgic et al. [53] introduced an adaptive real-time ON/OFF calibration with three adaptive delay compensating comparators that were able to achieve 82%-91% PCE and 92% VCE. In a conventional self-body bias circuit-based rectifier, the transistor parasitic capacitance may couple the ac signal from source to the body, bringing down the overall VCE and PCE. Substituting the self-body bias circuit with an adaptive body bias circuit, Cheng et al. were able to achieve PCE up to \sim 92% and VCE up to 96%, the highest values reported to date. A quadvoltage rectifier suitable for both low and high compliance voltages has been proposed in [17] and [56]. The proposed architecture consists of both HV and LV half-wave active rectifiers with a mixed-voltage gate controller to avoid substrate leakage current at high compliance voltage. This scheme is particularly suitable for implantable applications (e.g., neurostimulator implant) that require a high compliance voltage of around ± 12 V to provide biphasic current to the target tissues besides its low-voltage requirements ($\sim \pm 1.8$ V) for the digital controller and data telemetry circuits.

The high voltage (\sim 14 V) and high-power requirements (\sim 30 W) of LVAD or TAH impose particular challenges in designing the rectifier circuits, especially when the implants are powered by IPT at a high operating frequency [21], [67]. The challenges are mostly related to switching losses, power and volume constraints, propagation delays in the digital control path, the gate drivers and the switches, etc. The self-driven eGaN FETs in a synchronous rectifier configuration [see Fig. 3(g)] could be an optimal choice for these applications due to their high switching speeds, low ON-state resistance, and specifically low total gate charge, as reported in [19]. Employing this scheme, more than 95% PCE at 800 kHz operating frequency and \sim 30 W of output power was reported in this study.

C. Comparison of Rectifiers in Implant Application

A comprehensive summary of the comparative performances of the published state-of-the-art passive and active rectifiers used in biomedical implant applications is presented in Table II. It is quite evident that passive rectification is usually preferred for low-voltage (less than ~ 4 V) and ultralow-power implantable applications, such as cardiac pacemakers, vascular stent system, bone growth stimulator, and monitoring and recording system for targeted biological activities. In contrast, active rectifiers are not preferred to be used in ultralow and low-power (<1 mW) applications due to their additional power consumptions in associated comparators and MOSFET drivers. Therefore, if proper $V_{\rm th}$ mitigation technique and/or low-voltage threshold process are adopted, MOSFET-based passive rectifiers can offer superior performance compared to active rectifiers at lower power levels (typically less than 1 mW), which is in accordance with the findings of Szarka et al. [23] and Lehmann and Moghe [59]. However, some functional electrical stimulator implants (e.g., retinal, neural, and cochlear implants) and TAH/LVAD implants need output compliance voltage of 10-35 V and/or consume power in the range of milliwatt to a maximum of 30 W which, if the passive rectifier is used, may impose challenges in terms of high leakage current and substantial heat generation. In such cases, active or synchronous full-wave rectifier outperforms their passive counterpart, in terms of both PCE and VCE, at relatively high-voltage high-power implant applications. However, wide variations in input (alternating) voltage due to several factors such as human motion (vibration-based harvester), environmental variations (solar, TE generator), and coil misalignments (IPT) may affect the rectifier performance in terms of PCE, cold-start problem, etc. Adaptive reconfigurable rectifier proposed in [68] and [69] can address this issue by automatically changing its operating mode to either a voltage doubler or a rectifier circuit, depending on which one is a better choice for generating the desired output voltage at the highest possible PCE.

IV. AC-DC CONVERSION WITH VOLTAGE CONDITIONING

As mentioned earlier, the output voltage of the power management interface circuitry must comply with the load specifications. The required rectification, conversion, and regulation can be performed by one of the two stages: either by single-stage circuit discussed in this section or by separate rectifier followed by the dc-dc converter circuit reviewed in Section V.

A. Self-Commutated Capacitor-Based Converters/VM

Self-commutated capacitor-based converters can simultaneously provide rectification and voltage multiplication and are often referred to as VM. VM circuits are typically low-cost designs, built for basic voltage conditioning and do not regulate the

TABLE II SUMMARY AND COMPARISON OF THE STATE-OF-THE-ART RECTIFIER CIRCUITS USED IN BIOMEDICAL APPLICATIONS

Type	Ref	Output voltage (V)	Input signal frequency (MHz)	Power delivered (mW)	Load resistance (kΩ)	Maximum VCE (%)	Maximum PCE (%)	Target implant
	[38] ^a	1.2-1.8	13.56	0.005 ^b	20	-	66	Vascular graft monitor system
Passive	[16]	1.8	13.56	0.012	-	-	54.9	Body sensor network (BSN)
	[14]	2.5	0.256	0.02	-	80.2	-	Cardiac pacemaker
	[39] ^a	0.5-0.84	1100	0.35	2	-	34	Coronary stent system
	[66]	0.2	1860	0.5	0.1	-	55	Drug delivery system
	[46]	2.3-3	13.56	0.632	10	70	70-87	Cardiac pacemaker (stimulation)
	[50]°	0.39-	13.56	13.72	2	88.6	87	Biomedical implants ^d
		2.92						
	[49]	2.0	10	-	2	78	80	Biomedical implants ^d
Active	[59]	2.88	5	4.15	2	90	80	A/V prosthesis (Stimulations) e
	[57]	1.2-3.56	13.56	24.8	0.5-1.8	89-93	82.2-90.1	Biomedical implants ^d
	[58]	1-2.3	1.5	~ 50	0.1-2	84-95	87	A/V prosthesis (Stimulations) ^e
	[60]	2.5	13.56	~ 50	0.1	82	84.82	Epileptic seizure (Stimulations) ^e
	[55]	2-3.46	13.56	89	0.1-1	86-96	88-92	Biomedical implants ^d
	[56]	1.8 (LV)	2	10 (LV)	2.3 (LV)	82 (LV)	82-84	Retinal Implants (stimulation)
		12 (HV)		99 (HV)	5.4 (HV)	83.2 (HV)		
	[53]	2.4	10	-	-	92	82-91	Neural implant (stimulation)
	[54]	1.3	13.56	-	1	89	81.9	Biomedical implants ^d
	[61]	4.5-18	1.314	-	-	92	71.4	Neural Implant (Stimulations)

Note: References are sorted in ascending order based on their delivered power levels.

^a[38] uses 8 (eight) rectifiers in series. [39] uses 2 (two) rectifiers in series.

^bConsumed power.

^cResults are based on simulation.

^dDid not mention any specific implant.

eA/V prosthesis: audio/visual prosthesis

output dc voltage very well. They are fabricated from networks of capacitors, diodes, and switches that can be configured in a number of ways so that the dc output voltage is equal to some odd or even multiples of the peak value of the ac input voltage. Compared to full-wave rectifiers, the VM circuits offer better system efficiency as the maximum achievable power is higher due to the power being extracted at a higher voltage and a lower current, resulting in reduced conduction losses [70].

1) Passive VM: Passive VM circuits provide voltage level shifting at the expense of reduced PCE [23]. Most of the reported VM architectures use either junction-based diodes or other low-voltage drop devices (e.g., BJT, diode-connected MOS-FETs, etc.) [71]–[73]. A simple voltage quadrupler circuit is shown in Fig. 4(a), consisting of three junction-based diodes and three capacitors connected in a star configuration [23]. Torah *et al.* reported efficiencies in the range of 65% at 23 μ W load for an optimized Schottky-diode voltage quadrupler [72]. Interestingly, optimized Schottky diode-based VM can attain efficiencies 10% higher than the efficiency of those constructed from diode-connected MOSFETs due to the lower forward voltage threshold of the diode and additional leakage current resulting from the diode-connected MOSFETs not turning OFF completely. However, difficulty in the CMOS process fabrication of these junction-based diodes has instigated researchers to employ either diode-connected MOSFETs or active diodes in low-power biomedical applications [73]-[75]. In [73], a passive push-pull type voltage doubler, as shown in Fig. 4 (b), is implemented using diode-connected MOSFETs to power the auxiliary circuits of the implant IC. The authors reported that such devices should consume as low as 2.1% of the overall power (85 μ W) consumed by the IC. Although the diode-connected MOSFETs incur more forward voltage drop compared to active diode because of their threshold voltages, they do not significantly impact the overall PCE of the VM circuits given that the load current is small.

2) Active VM: Active diode-based VM can improve the PCE significantly due to the mitigated losses in its associated switches. Lehmann and Moghe [59] reported 15% increased efficiency when active voltage doubler was incorporated in place of the passive doubler at 4 mW of output power. The principles discussed in the active rectifier sections are relevant here since active VM uses the comparator-controlled MOSFET (active diode) as its pass switch. Lee and Ghovanloo [74] proposed an active voltage doubler for functional electrical stimulators, which can provide 2.84 V dc output across 1 k Ω load from a 1.46 V peak ac input at 13.56 MHz. This active topology was able to achieve the highest PCE of 79% ever reported at this high frequency, thanks to its high-speed comparators equipped with external offset control functions that can compensate for both turn-ON and turn-OFF delays, as shown in Fig. 4(c).

3) Comparison of VMs in Implant Application: Implant applications that naturally require ultralow power, such as cardiac pacemaker, bone growth stimulator, monitoring, and recording system for targeted biological activities, usually incorporate passive VMs for rectification since they do not significantly impact the overall PCE in such low power applications. At low excitation frequency (sub-100 Hz), charge-up losses within voltage source–capacitor loops dominate over the MOSFET conduction losses, while at higher frequencies conduction losses become more dominant [23], [76]. Therefore, passive VMs are also preferred to their active counterpart at the low excitation frequency. In contrast, active VM is typically preferred when power is harvested from high-frequency sources (e.g., inductive/UET)



Fig. 4. Various VM circuits. (a) Voltage quadrupler circuit in star configuration [23]. (b) Push–pull voltage doubler described in [73]. (c) Active voltage doubler proposed in [74]. Here, N_2 and P_2 are start-up switches driven by a complementary pair of start-up signals $S_{\rm UB}$ and $S_{\rm U}$.

to the implant. In addition, active VM can outperform passive VM to power relatively high-voltage, high-power biomedical implants such as neurostimulators, TAH/LVAD, etc.

4) Harvester-Specific Interactions: VM circuit designed for piezoelectric (PE) generators can use the piezoelement's internal capacitance as part of the circuit. However, VM is unable to apply complex conjugate matching to compensate for the relatively large capacitance in the PE transducer in order to extract maximum power from it [77]. Extremely low-voltage ($\sim 0.1-1$ V) and high-current output of EM generators may result in poor PCE in VM circuits (as well as in bridge rectifiers) due to the losses in diodes or diode-tied MOSFETs. Moreover, the bridge rectifiers and VM circuits behave as nonlinear loads making the design extremely difficult to track the maximum power from the energy harvesters [78], [79]. To overcome these issues, timed switching elements in ac-dc converter circuits for kinetic energy harvesters are proposed in the literature and discussed in the following section(s).

B. Direct AC-DC Switch-Mode Converters

Direct ac-dc switch-mode converter does not require a separate rectifier stage and usually incorporates switched inductors to pave the way for extracting maximum energy from the kinetic energy sources besides their regular ac-dc conversion. They can operate with very low input voltages and offer more effective current and voltage regulation than VM topologies [80]. The power profiles of the kinetic energy harvesters vary with human motion, environmental variations, etc., which implies that there exists a maximum power output point. These kinetic energy-based



Fig. 5. EM harvester-specific single-stage switched-mode ac-dc converter topologies. (a) Single inductor with split capacitor [81]. (b) Single inductor with secondary-side switches [78]. (c) Dual boost converter [80]. (d) Combined boost and buck-boost converter [79]. (e) AC-DC chopper combined with an NVC [83].

microgenerators are generally spring–mass–damper- based systems, whose damper characteristics (or damping force) need to be optimized to extract maximum energy for powering potential IMDs. The preferred way to achieve this optimal damping force is to adjust the effective output impedance of the EM and PE transducers or electric field between the two electrodes of the ES transducer with the help of direct ac–dc switch-mode power converters [80]. In light of the aforementioned discussion, it is evident that direct ac–dc power converters can extract maximum power from kinetic energy harvesters besides regular voltage conditioning.

1) EM Harvester-Specific Switched-Mode AC–DC Converters: Fig. 5 illustrates the direct ac–dc boost converter topologies proposed in [78] and [80]–[83], and most of them exploit the EM generators as input sources. These converters operate in discontinuous conduction mode (DCM) to reduce the inductor size requirements and MOSFET's power dissipation, making them suitable for exploiting low-voltage low-power EM harvester. Converter topologies shown in Fig. 5(a) and (b) are single inductor topologies, where the dc bus relies on two split capacitors in circuit (a) [81] and two additional MOSFET switches referred to as secondary-side switches in circuit (b) [78]. Both converters produce a regulated dc output voltage of 3.3 V from an ac input voltage range of 0.2-0.8 V and achieve conversion efficiency in the range of 60% at an input power level of around 100 mW. However, the split-capacitor architecture is simpler since it uses a reduced number of switching devices and requires no additional sensors to detect the polarity of the input voltage. In contrast, direct ac-dc converters equipped with two-stage regulation and input polarity sensors have been proposed in [79] and [80] and are shown in Fig. 5(c) and (d). The positive and negative input voltage cycles are processed by two boost converter stages in circuit (c) [80] and a buck–boost and a boost converter in circuit (d) [79]. The efficiency of the dual boost converter is reported as 50% with an input power of 50 mW, whereas the combination of boost and buck-boost converters achieves an efficiency of 61% with an input power of 100 mW. Since the output dc bus of the dual boost converter is split into two series-connected capacitors that suffer from large voltage drop when connected to the load, it is less efficient compared to the converter comprised of boost and buck-boost topologies. To achieve high conversion efficiency, Bolt et al. proposed an ac-dc converter that implements an ac-dc chopper combined with a negative voltage converter (NVC) consisting of rectifying cross-coupled MOSFETs connected in a bridge configuration [see Fig. 5(e)] [83]. Instead of using an external switched inductor, this architecture exploits an EM generator's internal inductance. This converter achieved 88%-90% PCE with a quiescent current of 250 nA.

2) ES Harvester-Specific Switched-Mode AC–DC Converters: It is worth noting that, unlike EM and PE generators, the damping force, being voltage controlled, can be dynamically optimized for ES generators. However, one of the main challenges for designing a converter for an ES generator is to minimize the parasitic capacitance connected to the generator, which will otherwise affect the overall converter efficiency [80]. In addition, a significant amount of voltage down-conversion is needed for powering low-power implant loads because of the high voltage (10-220 V) generated by the ES generator. In [84], a power processing circuit for a constant-charge ES generator was proposed. The design of the power processing circuit includes a buck converter to regulate the high output voltage from the ES generator to suit the power demand of the implant electronics. However, as the generator voltage rises, the depletion layer capacitance of the blocking junction of the high-side MOSFET forms a parasitic capacitance in parallel with the generator capacitor, consequently further reducing the converter efficiency. Mitcheson et al. presented a modified fly-back converter as a viable candidate for the power processing of biomedical implants while addressing the above-mentioned issues [80].

3) Piezoelectric (PE) Harvester-Specific Switched-Mode AC–DC Converters: The output impedance of a PE harvester is dominated by capacitive reactance, which becomes prohibitively large at the low operating frequency of such harvester. Therefore, the optimum load for maximum power transfer would require a significantly large inductance, possibly in the order of several tens to hundreds of Henry [23]. A possible solution to the abovementioned problem is to enhance the conversion efficiency by flipping/reversing the PE voltage at zero-crossing instants of the velocity of the proof mass (i.e., electrical current), as depicted in



Fig. 6. Voltage flipping solution by implementing the SSD method [85], [86]. Here, t_{Fi} corresponds to the time instants when piezovoltage is induced by the switching process on zero speed values (i = 1, 2, ...).



Fig. 7. Different SSHI topologies for PE energy harvesting. (a) Series-SSHI topology [23]. (b) Parallel-SSHI topology [23]. (c) Optimized series-SSHI topology [89]. (d) SSHI-MR topology [90]. (e) Series-SSHI with pulsed energy feedback [92].

Fig. 6 [85], resulting in maximum generated voltage, which exceeds the peak open-circuit voltage. The voltage flipping allows the circuit to maximize the harvested energy by extracting stored charge within the internal capacitances of the PE harvester at the extremes of the displacement cycle; otherwise, this energy would become waste. This voltage flipping can be implemented in a way, namely the synchronized switch damping (SSD) method, where a resonant electrical network is formed by placing an inductor between the harvester device and the rectifier. The combined PE capacitance ($C_{\rm P}$) coupled with the rest of the harvesting circuit acts as a tunable resistor ($R_{\rm CONV}$), as shown in Fig. 6 [86].

a) Synchronized Switch Harvesting on Inductor (SSHI): Voltage flipping using the SSD method can easily be realized by the *LC* resonant loop formed in the SSHI topologies, as shown in Fig. 7 [88]. Such SSHI topologies achieve an increase in the power gain above 4–15 times compared to direct charging



Fig. 8. (a) SSHI circuit incorporating active diode-inserted-resonant-loop [77] and (b) PZT internal capacitor voltage. Here, t_F = the optimal flipping time and V_F = the flipping voltage = V_R - V_D , V_D = diode voltage drop. The additional active diode in each resonant loop holds the capacitor voltage upon reaching its peak in the opposite polarity at t_F .

through bridge rectifiers only [87], [88]. The most common forms of SSHI circuits are series-SSHI and parallel-SSHI, where series-SSHI uses an inductor in series with a rectifier, whereas a parallel-SSHI configuration uses a parallel inductor, as depicted in Fig. 7(a) and (b), respectively. Optimized series-SSHI shown in Fig. 7(c) provides further efficiency improvement by replacing four diodes of the bridge rectifier with two diodes and a selector switch [89]. The authors reported an increase in harvested energy output as much as 120% of the parallel-SSHI circuit. Garbuio et al. proposed a series-SSHI circuit with a transformer called SSHI using magnetic rectifier (SSHI-MR), and the topology is shown in Fig. 7(d) [90]. This SSHI-MR can facilitate start-up and, most importantly, can maintain good efficiency at a very low input voltage level due to having a lower number of energy dissipating discrete components (i.e., diodes). Lallart et al. presented a rectifier that can automatically switch between SSHI-MR and parallel-SSHI topology according to the harvester voltage levels [91]. Lallart and Guyomar proposed a series-SSHI with pulsed energy feedback to the PE element [92], as shown in Fig. 7(e), and achieved ten times power gain compared to the conventional SSHI technique presented in [93]. To increase the efficiency of PE energy harvesting over a wide operation range, integration of SSHI and maximum power point tracking (MPPT) is proposed in [94].

A significant amount of research is dedicated to overcoming the design challenges of SSHI that lie in the precise timing control to switch OFF at the peak voltage (voltage flipping) with minimal circuit complexity of the controller. This feature is critical since the efficiency of the circuit is sensitive to the flipping time [77]. For instance, in order to achieve precise timing control, Sanchez et al. [95] used an external variable resistor, whereas Ramadass and Chandrakasan [96] introduced an external 8-bit digital module, resulting in 4-5.8 times more power extraction compared to the state-of-the-art standard FBR or voltage doubler with the same displacement amplitude. To avoid external tuning circuits and their associated power consumption, Aktakka et al. [97] and Lu and Boussaid [98] proposed a parallel-SSHI topology where an additional p-n junction diode is inserted in the resonant loop followed by an FBR. The additional diode enforces a unidirectional flow of the loop current, which maintains the capacitor voltage upon reaching its peak in the opposite polarity, thereby eliminating the need to tune the switch OFF at the precise time $(t_{\rm F})$ of capacitor peak voltage, as shown in Fig. 8(b). Unlike [97], [98], the design approach proposed in [77] inserts an active diode (instead of a p-n junction diode) for each resonant loop and does not require any subcircuit



Fig. 9. (a) Structure of a power converter stage based on SECE topology for piezoelectric transducers. (b) Boost topology for SECE. (c) Flyback topology for SECE [99].

for polarity detection, which further simplifies its controller operation [see Fig. 8(a)]. This active diode-inserted-resonant-loop approach achieves a maximum PCE of 85% and output power of 136 μ W in addition to its 210% improvement in the maximum power extraction compared to a conventional rectifier FBR.

b) Synchronous Electrical Charge Extraction (SECE): One of the main drawbacks of the SSHI technique is that the efficiency is bias-dependent and quite low in extreme load conditions [99]. The SECE method is another well-known energy extraction technique for PE harvesters where the generated power is almost load-independent [99]-[108]. Unlike the SSHI technique, in SECE, energy is first transferred from the transducer to an inductor upon detection of the peak voltage. This energy translates to the output through two resonant circuits after that, in a way similar to a boost converter, as shown in Fig. 9 [99]. This unique topology makes the efficiency of an SECE independent of the output bias, while the complete removal of electrical charge at each activation doubles the peak voltage. Lefeuvre et al. demonstrated that the SECE circuit achieved a power improvement of 400% compared to the FBR technique [100]. In order to reduce losses on freewheeling diodes, SECE can also be implemented with a flyback topology [107]. To achieve better performance and simplify the enhanced complexity of the control and switching mechanism associated with SECE, several enhancements were made on the basic SECE topology, such as the optimized SECE (OSECE) by Wu et al. [101], adaptive pulsed SECE by Hehn et al. [108], the self-powered OSECE by Liu et al. [102], SECE with residual charge inversion by Dini et al. [99], and SECE with rectifier-free, bidirectional converter topology by Shareef et al. [103]. Nevertheless, these SECE rectifiers are not suitable for strongly-coupled PE harvesters. For designing an SECE converter in a strongly-coupled PE environment, some guidelines are provided in the tunable SECE by Lefeuvre et al. [104], the synchronous electric charge partial extraction by Xia et al. [105], and short-circuit synchronous electric charge extraction based on a short-circuit tunable phase by Morel et al. [106].

c) Comparison and Implementation of SSD Methods in Implant Applications: Parallel and series-SSHI methods may exhibit outstanding performances under periodic vibrations

Fig. 10. Schematic of a conventional LDO regulator [117].

from internal organs (lung and cardiac motions, blood circulations, etc.). However, in most of today's implantable applications, kinetic energy is extracted from the movement of skeletal muscle under voluntary control, where the vibrations are not periodic and mechanical stimuli occur at unpredictable rates. The SSHI scheme presents inherent weaknesses of poor impedance matching and voltage regulation while harvesting such sporadic vibrations (i.e., variable amplitude) [109]. In contrast, the SECE scheme outperforms its SSHI counterparts in terms of power gain while harvesting energy from weakly coupled PE harvester systems or PE harvester systems vibrating under OFF-resonant conditions due to the unpredictable mechanical vibrations of the skeletal muscles [110]. However, unlike SSHI methods, the SECE scheme may demonstrate poor performance at harvesting PE energy for ultralow and low power implants because of the power consumption in their complex internal switching and control circuits.

The above-mentioned state-of-the-art direct ac-dc switchmode converters for kinetic energy harvester require no less than one switched inductor to facilitate the maximum power extraction. The power converters using off-chip inductors not only require more board space but also pose substantial health risks due to the susceptibility of the inductor to EMI [111]. However, since maximum power extraction is of primary importance while harvesting ambient energy sources for powering biomedical implants, converter design and optimization should be carefully accomplished to minimize the size and number of the switched inductor.

V. DC-DC CONVERSION WITH VOLTAGE CONDITIONING

An alternate option to using a direct ac-dc voltage conversion technique is to use a separate rectifier followed by a dc-dc converter circuit. While the rectifier stage can be realized by any of the technologies described in the previous sections, dc-dc converter topologies will be discussed in the following sections.

A. LDO Linear Regulators

LDO regulator provides a stable stepped-down dc output voltage independent of load impedance, input-voltage variations, temperature, and time [112]. Fig. 10 shows the typical configuration of an LDO regulator. An LDO usually comprises three basic functional elements: a pass element (either a BJT or a MOSFET), a reference voltage, and an error amplifier. Typically, the pass element governs the accuracy of the output voltage by being always ON and dissipating continuous power. The error amplifier senses the output voltage continuously, compares it with the reference voltage, and generates a compensated control signal that changes the pass element's voltage drop to maintain a constant output load voltage.

LDO can be classified into two categories based on the presence of an output capacitor in the design: LDO with capacitor [113], [114] or LDO without capacitor [115], [116]. LDO with capacitor has the benefit of simpler circuit architecture to maintain adequate stability over a wide range of loading conditions. However, the relatively large-sized output capacitors in the single microfarad (μ F) range required for this type of LDO cannot be realized in the system-on-chip (SoC) approach of the implantable devices. Therefore, an off-chip external capacitor is needed for this type of LDO to maintain stability under all operating conditions [117]. Nevertheless, the equivalent series resistance of the external capacitor reinforces the ripple in the output voltage during load transients [118]. In contrast, capacitorless LDO can reduce the size of the implants significantly since it does not require an external off-chip capacitor. For example, Chen et al. proposed a capacitorless LDO design with a high power supply rejection ratio (PSRR) for artificial retina application that can provide a stable output voltage of 3.3 V under the output current variation of 1-5 mA while achieving a PSRR of -46 dB at 10 MHz [119]. However, a robust compensation scheme is required for this kind of LDO to maintain the transient response stability and alternating current stability that otherwise would have been carried out using the presence of an external output capacitor [120].

The biomedical implant systems may require multiple independent regulated voltage sources to power different submodules. In a study presented in [121], a multiple-output LDO (MOLDO) with an off-chip external capacitor has been proposed, where one MOSFET is time-shared by four output legs to produce four individual outputs. These four output terminals are configured with different voltage ratings with a maximum single output voltage of 3 V from an input voltage of 3.3 V. Although the proposed MOLDO can minimize power consumption by providing independent regulated voltage for each submodules of an implant system, it suffers from slow settling time and large ripples in the output voltages arising from a small feedback factor. To overcome this limitation, Mo et al. introduced a constant feedback factor-based time-shared technique in MOLDO architecture that reduces the ripples in the output voltages by enhancing the settling time through constant feedback [122].

The frequency stability of the above-mentioned analog LDOs (ALDOs) is highly load-dependent [118]. To address this problem, the error amplifier of the ALDO is replaced by a digital control logic circuit in a digital LDO (DLDO) architecture. In a DLDO, the output voltage is digitized by an analog-todigital (ADC) converter and fed to a digital controller [123]. A compensation digital code is generated by the digital controller and supplied to a digital-to-analog (DAC) converter to produce the required output voltage. DLDO offers the advantages of a wide range of load variations, lower sensitivity to process variations, scalability, size, and low voltage operation [118], [123]. For instance, A DLDO has been reported in [124] that works at an input voltage as low as 0.5 V at a current efficiency of 98.1%, thereby making it suitable for powering ultralowpower implant electronics. However, DLDO can suffer from load current-dependent ripple in steady-state condition due to digital loop compensation in discrete steps [118]. Ripple cancelation amplifier can minimize the ripple in the output at the expense of increased circuit complexity.

The quiescent current consumed by the series pass transistor of an LDO in its idle stage contributes to reduced efficiency. Continued research efforts have scaled down the quiescent current consumed by an LDO to a value as low as 0. 062 μ A, thereby minimizing power loss [125]. To date, LDOs have been designed to power a wide range of implantable devices operating at a voltage as low as 0.9 V [126] to a voltage as high as 10 V [127].

LDOs are an efficient choice over the switched-mode converters when the current is relatively low or the voltage drop is relatively small. Therefore, LDOs have been extensively used in the power management units (PMUs) of ultralow and low power implantable applications such as wireless intracranial pressure monitoring system [128], artificial retina application [119], spinal cord stimulator [120], neural stimulator [17], [56], blood flow sensor microsystem [38], cardiac microstimulator [14], total knee replacement (TKR) [129], and many more. Other attractive features of LDOs with respect to implantable applications are low-noise output voltage, small size due to fewer external components, low shutdown current, and low cost. In fact, the clean voltage generated by the LDO is an ideal choice for noise-sensitive, high-frequency analog and digital circuits such as phase-locked loops for microprocessors, high-speed communication links, etc., inside PMUs of IMDs. However, voltage up-conversion is not possible by LDO which limits their applications in harvester-driven IMDs.

B. DC-DC Switched-Inductor Converters

A large and growing body of literature has investigated the classic switch-mode inductor-based converter topologies to power IMDs from ambient energy sources. A step-up converter with a cold-start mechanism is needed when considering powering IMDs from extremely low-voltage microenergy sources, such as thermoelectric energy generator (TEG) devices, MFCs, and solar PV energy sources. Moreover, the voltage produced from the kinetic energy harvester needs to be stepped up or down depending on the requirements of the implant electronics. Several studies thus far have evaluated dc–dc switched-inductor converters for harvesting energy from TEG devices [99], [130]– [140], MFCs [140]–[142], and solar PV energy sources [133], [143]. In addition, numerous studies have attempted to focus on combining energy from multiple sources to further improve system reliability and increase power availability [144]–[146].

1) Single-Input Single-Output (SISO) Topology: The most widely used switched-inductor converters are realized on a single output utilizing a single inductor in its direct power transfer path to the load [99], [131]–[136], [140]–[142]. Many of these converters introduce hybrid-type architecture including a separate dedicated circuit for MPPT, cold-start (start-up), additional auxiliary converters, etc., to achieve the requirements for low



Fig. 11. Single-stage SISO step-up converter as a TEG energy harvesting interface proposed in [130].



Fig. 12. Cascade-type two-stage SISO switched-inductor dc–dc converter as a TEG energy harvesting interface proposed in [139].

input voltage and high conversion efficiency simultaneously. For instance, Weng et al. presented a fully electrical start-up, a batteryless step-up converter that can generate a regulated 1.2 V output voltage from a TEG with a minimum input voltage of 50 mV and maximum efficiency of 73% [130]. The schematic is shown in Fig. 11 that consists of four functional blocks: a low-voltage start-up circuit, an auxiliary step-up converter, a ZCS-controlled boost converter and peripheral controllers. The auxiliary step-up converter acts as an intermediate buffer to prevent the undesirable loading effects during the cold start period. A two-stage topology with a cascaded auxiliary boost converter and dc-dc buck converter for a TE energy harvesting system has been proposed by Ramadass and Chandrakasan [139] (see Fig. 12). This dc-dc converter can operate down to an input voltage of 25 mV and provide a regulated output of 1.8 V with an overall end-to-end efficiency of 58%. A human motion-activated switch-based boost converter acts as a cold-start circuit. The auxiliary boost converter processes and transfers energy from the harvester to the storage capacitor ($C_{\rm STO}$). The dc–dc buck converter starts powering the load at regulated voltage $1.8 V(V_L)$ when the storage capacitor voltage ($V_{\rm STO}$) reaches 2.4 V. All these SISO topologies obtain reasonable efficiencies at relatively high input power (mW) levels. However, the efficiencies degrade as the input power decreases and become insufficient at μW power levels. This is due to the fact that the control circuit is usually powered by the converter output whose voltage level is determined by the requirements of the implant load rather than optimized for the low power consumption of the control



Fig. 13. Single inductor SIMO step-up dc-dc converter topology based on AGB and ERC techniques proposed by Chen *et al.* [137].

circuit. As described in the following sections, a multiple-output architecture is employed to achieve a high conversion efficiency at μ W input power levels by reducing the power consumption of the control circuit to nanowatt level [138].

2) Single-Input Multiple-Output (SIMO) Topology: Timesharing of a single inductor in DCM allows the switchedinductor converter to add load-independent output(s) that ensures high efficiency over a wide range of power for multiple reasons. First, the additional load-independent output can be optimized for the low power consumption of the control circuit, and second, to start up the converter, instead of charging a large storage load capacitor, a much smaller additional output capacitor(s) can be charged. Last but not the least, by providing multiple outputs through a single inductor, the SIMO converters can meet multiple voltage requirements of the low-power implant electronics with minimal dynamic power consumption and enable a considerable saving in cost, weight, and size with minimum risk of EMI [147]. Chen et al. proposed a single inductor SIMO step-up converter for TE energy harvesting with adaptive gate biasing (AGB) and near-threshold voltage energy redistribution control (ERC) technique that can optimize power consumption over a wide range [137]. The schematic of the converter architecture is shown in Fig. 13. The AGB technique produces two load-depended gate driving voltages (V_{AGB1} and V_{AGB2}) for the power switches of the primary boost converter in such a way that it reduces conduction and switching losses under heavy-load and light-load conditions, respectively. The proposed converter can boost an input voltage of 100-500 mV at $25-100 \,\mu\text{W}$ of output power and achieved a maximum efficiency of 83.4%, which is the highest efficiency reported to date for a boost converter at this low output voltage level. However, the efficiency of the converter drops drastically beyond the output voltage of 500 mV because of the additional switches in the AGB block along with the large inductor (1 mH) that incur a significant loss at high voltage stress. Another single-input dual-output (SIDO) boost converter is shown in Fig. 14, which together with its ZCS and ZVS switching techniques and additional loadindependent output ensures high conversion efficiency at very low input power (microwatt) levels [138]. The boost converter can start working at an input voltage as low as 15 mV and the control circuit consumes only 160 nW of quiescent power. It



Fig. 14. Single inductor SIDO step-up dc-dc converter topology for harvesting TEG energy proposed by Katic *et al.* [138].



Fig. 15. Single inductor dual-input dual-output (DIDO) boost converter topology proposed by Katic *et al.* [144].

can boost a thermal input voltage of 45 mV to a regulated output voltage of 1.9 V with a peak conversion efficiency of 86.6% at 30 μ W input power.

3) DC-DC Converters for Multisource Energy Harvesting Interface: All the previous switched-inductor converter architectures rely on a single source to power the load electronics. However, microenergy harvesting for biomedical implants from a single source may often experience issues such as inadequate output power and poor reliability due to its unpredictable nature. In contrast, energy harvesting from multiple sources can improve the overall system reliability, robustness, and can increase the attainable output power. Several attempts have been made to investigate multisource platforms that combine the power from GBFC-TEG [144], PV-TEG-piezo [145], PV-RF-piezo [146], and so on. Combining energy from these multiple sources is facilitated by inductor sharing using DCM operation where the inductor is alternatively time-shared with one harvester at a time to deliver power to the output load while the switches correspond to the other harvesters are kept OFF. For instance, Katic et al. presented a single-inductor dual-output boost converter topology that combines energy from implanted GBFC and TEG devices, and the architecture is shown in Fig. 15 [144]. The control block is powered from the second output (V_{CTRL}) and the digital signal $V_{\rm ST_OK}$ controls the switch that connects the load to the main output provided that enough energy is accumulated in the storage capacitor ($C_{\rm ST}$). The low quiescent power consumption of the control block together with both ZCS and ZVS switching of the converter enable it to achieve a peak conversion efficiency of 89.5% at 66 μ W of combined power delivered by the two harvesters. The converter obtains a peak power efficiency of 85.32% at an output power of 23 μ W for the TEG source and 90.4% at 29 μ W for the GBFC source. Similarly, Bandyopadhyay and Chandrakasan [145] proposed a



Fig. 16. Inductor sharing scheme in a single inductor MIMO switching converter [148].

multisource platform that combines power from solar, thermal, and PE sources. A boost converter is employed for thermal and PV inputs and a rectifier followed by a buck–boost converter is implemented for the PE input. The system can boost the input voltages from 20 mV to 5 V and achieved a power efficiency in the range of 58% to 83% when using individual harvesters and a peak power efficiency of 96% when considering overall end-to-end efficiency.

4) Inductor Sharing Scheme in Implant Applications: The above-mentioned multiple port dc-dc switched-inductor converters including SIMO, multi-input single-output (MISO), and multi-input multi-output (MIMO) power conditioning architectures allow combining multiple converters into a shared inductor converter by conducting time-sharing of the inductor utilizing the idle time of the DCM operation [147]. Fig. 16 shows a schematic of an exemplary inductor sharing scheme in a singleinductor MIMO buck switching converter [148]. This suffices the need for separate inductors for either additional outputs or inputs in these converter topologies, resulting in significant savings in cost and area for space-constrained implant applications. Reducing the number of required inductors in switched-inductor converters also mitigates the EMI-related health risk. However, at an ultralow-power implant system, the complex control circuit overhead associated with inductor sharing limits the system's overall efficiency. In order to achieve better efficiency in such ultralow-power ranges, greater attention to design optimization is needed that may include power gating, dynamic pulse width control, ZCS and ZVS techniques, and so on [138], [143].

C. DC–DC SC Converters

DC–DC SC converters or charge pumps (CPs) use capacitive energy transfer instead of inductive energy transfer associated with dc–dc switched-inductor converters. These converters are combinations of switches and capacitors where the charge is transferred from one capacitor to another under the control of regulator and switching circuitry [149]. In addition to being more immune to EMI, the inductorless design of SC converters can lead to a drastic reduction in the complexity and cost of the on-chip integration and fabrication process, and consequently, are widely employed in implant applications [150]–[152].

Among various SC topologies, the Dickson, voltage doubler, series–parallel, and Fibonacci topologies are widely used in implantable applications [6], [7], [15], [16], [73], [153], [154].



Fig. 17. Schematics of different SC topologies. (a) Conventional Dickson CP [155]. (b) Modified Dickson CP used in [6]. (c) Pelliconi architecture based three-stage negative charge pump used in [16]. (d) Fibonacci 5:1 SC network used in [154]. (e) Series–parallel SC topology used in [7].

Fig. 17(a) shows a basic structure of a Dickson CP topology where capacitors are interrelated by two nonoverlapping clocks and coupled in parallel with diode-connected MOSFETS [155]. However, the voltage pumping gain is limited by reversecharge-sharing effects and the MOSFET threshold voltage. To overcome these limitations, a modified Dickson CP is adopted, which utilizes additional parallel auxiliary switches as static charge transfer switches and voltage controller consisting of pass transistors [see Fig. 17(b)] [6], [153]. For example, Lee *et al.* designed such a modified CP for a cardiac pacemaker that can generate a stimulation voltage of 3.2 V from a 1.2 V dc supply [6]. Chen *et al.* [156] proposed a four-stage on-chip Dickson CP with parallel-connected photodiodes for subdermal implant applications. By employing an auxiliary charge pump using zero-threshold voltage (ZVT) devices in parallel with the main charge pump, this system obtained a low-voltage start-up of 0.25 V and an output power of 1.65 μ W at 1.06 V from a 0.31 V input.

With the Pelliconi charge pump design, dual-polarity can be achieved, thus making it suitable for biphasic stimulation with small area utilization [15], [16]. Pelliconi [157] CP uses antiphase clock signals to actively operate cross-coupled MOS switches along with two coupling capacitors. Each stage of the Pelliconi CP circuit acts as a voltage doubler, and like the Dickson charge pump circuit, the gain ratio (GR) can be enhanced by cascading multiple stages. For example, Ethier and Sawan proposed two separate charge pumps based on Pelliconi architecture to generate ± 9 V driving voltage from 3.3 V dc supply that facilitates biphasic intracortical current-pulse stimulation [16]. The schematic of the proposed three-stage negative charge pump intended to generate -9 V supply is shown in Fig. 17(c).

Fibonacci CP proposed by Makowski *et al.* is an *n*-stage charge pump where the voltage gain increases or decreases by the (n+1)th Fibonacci number as opposed to the Dickson or Pelliconi charge pump, where the voltage gain increases linearly [158]. A 5:1 Fibonacci CP, shown in Fig. 17(d), was reported by Wieckowski *et al.* for implantable applications that achieve output voltage one-fifth of input supply of 3.6 V while delivering 250 nW of power [154].

CP with series–parallel topology has the best switch-capacitor utilization among the other basic SC topologies. From an areacost point of view, it is the most suitable CP to be applied in implant PMUs. Fig. 17(e) shows a schematic of a basic 3X series–parallel SC topology used in a single-chip very-lowpower interface IC that delivers a high-voltage stimulation for cardiac pacemakers [7].

1) Regulation of SC Converters: Conventional SC converters suffer from poor efficiency in a wide range of $V_{\rm in}$ and/or $V_{\rm o}$ values [150]. However, the implant PMU must efficiently meet its varying load and different voltage requirements of different functional blocks, as well as various operating modes. Multiratio SC converters (MRSC) with reconfigurable power stages can provide a wide range of output voltages catering to these different voltage requirements, with the added benefit of minimal power consumption [151], [152].

a) Multiratio SC Converters: Several MRSC converter topologies, such as step-up [159], step-down [136], [152], [160]–[163], and both step-up and down converters [151], [164], [165], have been evaluated for ultralow power applications in the literature, where the power stage can be reconfigured into different architectures to achieve multiple GRs. One such reconfigurable SC converter power stage was reported by George et al., which is shown in Fig. 18 [151]. The converter can be reconfigured to achieve five GRs in both step-up and step-down conversion modes and can deliver up to 7.5 mW to the load with maximum 75% efficiency, using an active area of only 0.04 mm².

MRSC converter has lower than the expected efficiency due to its inability of providing low output impedance and increased



Fig. 18. Five-GR reconfigurable SC converter power stage along with its different switching schemes [151].

number of switches and control circuitries. The efficiency of the MRSC converter is noticeable at ultralow ($\sim 0.1 \mu$ W) to low power ($\sim 1 \text{ mW}$) implantable loads, such as a pacemaker, bone growth stimulator, body area network and cochlear implant [26], [150]. Low power density applications pave the way for operating the converters in relatively lower switching frequencies, thereby reducing the switching losses. Therefore, at this low power level, the efficiency of this MRSC converter becomes substantially dependent on the conduction losses that can be minimized by using increased GRs.

b) Control Strategies Applied in SC Converters: In order to maintain the line and load regulations of the SC converters, various control strategies have been implemented as of now. According to (1), four variables such as GR (M), switching frequency (f_{sw}), duty cycle (D), and switch conductance (G) can be changed to control the output voltage of an SC converter [150]. The output of the SC converter can be defined as

$$V_{\text{out}} = M V_{\text{in}} - I_{\text{out}} R_{\text{out}} \left(f_{\text{sw}}, D, G \right). \tag{1}$$

One of the most common control strategies used in implant applications is conversion ratio control, also known as gear-box control or adaptive gain (AG) control that adjusts the *M* to provide the required output voltage [151], [152], [160]–[162], [164]. Since conduction loss is directly affected by the number of conversion ratios, it is the only control method that can reduce the conduction loss [26]. The remaining three control variables, f_{sw} , *D* and *G*, modify the output impedance (R_{out}) of the converter for voltage regulation, conceptually similar to the lossy regulation of a series linear regulator. Out of these three control variables, only switching frequency (f_{sw}) can be easily varied over the necessary range [150]. Therefore, the efficiency of the SC converters in ultralow-power/low-power implant applications is primarily dependent on minimizing switching losses of this control strategy.

Traditionally, to achieve fine regulation, SC converters adopt PFM in combination with the AG control, as described in [151], [152], [162], and [164]. This is preferred for ultralow-power and low-power implant applications because PFM keeps switching losses proportional to load current, leading to better efficiencies. In general, two types of PFM controllers are reported in biomedical applications, namely, discrete or automatic frequency scaling (DFS) controller [151], [162], [164] and asynchronous controller [152]. The DFS controller scales the switching frequency in several steps to regulate the output voltage. For instance, George *et al.* [151] proposed a step-up/down SC converter using AG and DFS controller that is capable of regulating the output voltage within a range of 1-2.2 V derived from an input voltage of 1.8 V by scaling the frequency in four steps (1.25/2.5/5/10/20 MHz). It can deliver a load power of 7.5 mW with a reported efficiency of up to 75%. However, the DFS controller requires a constant running clock and frequency divider, thereby increasing the dynamic power consumption of the converter even if the load is not active. In contrast, the asynchronous controller, proposed in [152], is able to turn on the minimum number of switches while keeping the additional switches OFF depending on the load current (event-driven technique). Therefore, it can potentially reduce power consumption during low-load scenarios. PFM control scheme suffers from switching frequency variation with load and input voltage change, thus making it difficult to filter out the unwanted noise from the controlling signal. A delta-sigma closed-loop-based control technique may be useful in suppressing the noise by spreading the generated unwanted tones across a wider frequency spectrum [166].

In biomedical applications, several attempts have been made to carry out the series connection of an LDO voltage regulator with the SC converter [6], [73], [154], [161]. In this scheme, the SC converter is designed to make up most of the voltage gap between the inputs and the expected output voltages (coarse tune), whereas the LDO is designed to fine-tune the output voltage toward the desired target voltage. This hybrid type of converter is preferred when the input to output voltage ratio is high enough to achieve high efficiency using either SC converter or LDO alone. Other advantages of LDO-coupled SC converters include reduced noise at high-frequency operation, low input and/or output voltage ripple, and faster dynamics in the load regulation [26]. Wieckowski et al. [154] proposed a hybrid SC converter network consisting of a 5:1 Fibonacci CP cascaded with LDO. The Fibonacci SC converter alone provides 720 mV from a 3.6 V Li-ion battery source and it is further stepped down to 444 mV when connected in series with an LDO. This hybrid configuration achieves an output voltage ripple of less than 50 mV at 56% power efficiency (4.6 \times of an ideal linear regulator) under 5 μ W of loading conditions. However, the main drawback of this hybrid approach is that in order to allow for the voltage drop across the LDO, the $I_{out}R_{out}$ drop of the SC converter needs to be even lower, which implies the use of a higher switching frequency, larger switches, higher capacitors, or specific combination of them. Any of these would result in an increase in the switching losses.

c) Multiphase SC Converters: Unwanted noise generated in the SC converter may also be reduced by splitting it into smaller cells in a modular approach. This type of converter is called the multiphase SC converter that has been implemented in many applications reported in [159], [160], [162], [163], and [167]. While the SC converters suffer from large output voltage ripples and input inrush current because of their separate charge-discharge path, the interleaving technique used in multiphase converter minimizes these drawbacks. In the interleaving technique, a part of the converter's charge transfer capacitances can be shared among multiple-stage cells by time multiplexing the evenly distributed clock phases. In addition, the multiphase configuration may allow a significant reduction of the size of the output capacitance resulting in a significant reduction of the size



Fig. 19. Multiphase architecture of the SCPC proposed in [167].

of the converter chip, and thereby decreasing the size of the target biomedical implant as a whole. Fig. 19 shows the architecture of a multiphase SC converter proposed in [167]. The converter obtains two different GRs (1/2 and 2/3) with an efficiency of 81% at a power density of 38.6 mW/mm². As expected, the measured output voltage ripple is as low as 3.8 mV, which is less than the ripple produced by the single-phase SC converters having similar output capacitance.

VI. COMPARISONS OF DIFFERENT POWER CONVERTERS

A. Strengths and Limitations of Different Power Converters

A comprehensive analysis comparing the characteristics of different power converters will be presented in this section. Because the use of IMDs has increased significantly over the years, more research focus is being driven toward improving the efficiency of the power converters that are used as the interface between energy harvesters and implants. Therefore, linear regulators (e.g., LDOs), switched-capacitor power converters (SCPC), and switched-inductor power converters (SIPC) are being predominantly employed as the voltage conversion blocks to condition power for the IMDs [26].

LDO architecture is one of the widely used elements of power management for applications where volume and weight requirements are stringent with marginal voltage overhead [126]. Due to their ability to achieve noise-immune voltage regulation, LDOs are still popular in implant applications where noise isolation and emissions are major system concerns. That being said, in implant applications, the excessive energy dissipation across the LDO can impose a significant health risk, especially where the output voltage of an energy harvester is several orders higher than the load voltage. In addition, a large capacitor is required to stabilize the output of a conventional LDO regulator, which may become challenging in on-chip applications, especially without a substantial tradeoff in the form factor and chip area [117]. Capacitorless LDO regulators have been proposed in several studies with additional control circuits that increase the implementation cost and complexity of the regulator circuit [126], [168]. As a result, designers need to overcome significant design challenges to fully utilize the advantages of capacitorless LDOs. Moreover, LDO can only accomplish down-conversion of the input voltage



Fig. 20. Circuit implementations of (a) inductor-based and (b) SC-based stepdown converters showing key sources of efficiency losses [27].

rendering itself irrelevant in biomedical applications that require voltage up-conversion.

Switching power converters (i.e., SCPC and SIPC) can be configured to accomplish both voltage-up and voltage-down conversion depending on the biomedical application requirements. Moreover, these types of converters can achieve the required voltage conversion at a much higher efficiency (compared to the LDOs) in applications having a substantial mismatch between input and output voltages. The efficiency (η) of a switching converter can be expressed by the following equation [27]:

 $\eta =$

$$\frac{E_{\text{load}}}{E_{\text{load}+}E_{\text{conduction}+}E_{\text{switching}+}E_{\text{parasitics}+}E_{\text{control}+}E_{\text{leakage}}} \times 100\%.$$
 (2)

 \mathbf{T}

Here, $E_{\rm load}$ refers to the energy delivered to the load per cycle, and other terms in the denominator refer to different losses incurred during the operation of the switching converter. Simplified designs of the SC and switched-inductor-based converters are illustrated in Fig. 20. These designs are configured in a step-down mechanism, coupled with fundamental sources of efficiency losses.

SCPC operates based on the capacitive energy transfer mechanism. Since SCPC does not have an inductor, it is less affected by EMI and can be used in implants that utilize inductive links. For the same reason, they are ideal for integrated implementation. One of the major advantages of SCPC is the ability to start at lower input voltages when compared with other dc-dc converters, particularly SIPC. This advantage is often exploited during cold-start, which is detailed in Section VII. In some biomedical implant systems, a broad set of implementable conversion ratios may be required to support its different blocks or to meet the voltage differences between the input and the load that may vary substantially from time to time. Multiratio SCPC (MR-SCPC) instead of fixed GR SCPC can meet multiple voltage requirements with minimal dynamic power consumption by adopting dynamic voltage scaling (DVS). However, MR-SCPC suffers from heightened switching and conduction losses that can be attributed to the increased number of switches and capacitors incorporated to realize multiratio architecture. Conduction and switching losses of semiconductor switches and charge-up losses within source-capacitor loops are some of the most

prominent loss mechanisms of an SCPC [76]. At low switching frequency, charge-up losses dominate, while at higher switching frequency conduction losses and switching losses dominate [23]. Splitting the flying capacitor into smaller packages and activating the required number of packages depending on the required level of output power can become a viable solution to reduce the charge-up losses [169]. Switching losses within an SCPC can be minimized by opting for PFM at light-load conditions. One drawback of using PFM instead of PWM is the variable frequency associated with PFM that leads to a broad noise spectrum, thereby making the noise filtering process sophisticated and expensive. However, the noise content in the low switching frequency of PFM corresponding to ultralowpower/low-power conditions is less than its PWM-controlled counterpart. At higher switching frequency, PWM-controlled SCPC is more efficient compared to its PFM-controlled counterpart [26].

On-chip integration of capacitors using bulk CMOS technology and subsequent small form factor are two of the most prominent features of SCPC that fuel its widespread use in biomedical applications. Under the bulk CMOS technology, the capacitive density varies from 4 to 12 nF/mm², which may not be sufficient in many applications where large output current ripple suppression is needed [26]. Allocating a larger area to implement sufficient on-chip capacitance can solve the issue at the expense of increased implementation cost. Capacitors implemented through *metal–insulator–metal* and *silicon-on-insulator* technology can reduce the bottom-plate parasitic, which in turn can reduce the charge-up losses and increase the capacitive density [154], [170]. Capacitive density can reach up to a maximum of 400 nF/mm² by using trench capacitors, as mentioned in [171].

SIPC provides an alternative form of voltage conversion technology that utilizes a magnetic energy transfer mechanism to accomplish the required voltage conversion. Unlike SCPC, the efficiency of this type of converter topology is not limited by a fixed conversion ratio for a specific application, making them suitable for DVS implementation. Besides regular voltage conditioning, switched inductors in direct ac-dc power converters facilitate maximum power extraction from kinetic energy harvesters. The inductor of an SIPC can be integrated either monolithically or can be adopted in a system-in-package (SiP) approach to miniaturize the overall footprint. However, the current ripple stemming from the absence of adequate large inductance gives rise to a large rms current, inducing significant loss. Using ferrite as a substrate for chip-size solenoid inductor [172], incorporating off-chip SMD air-core inductors [173], and realizing reactive components and switches in different dies [174] are some of the SiP approaches that can incorporate the required reactive components at comparatively low cost. Nevertheless, monolithic integration of inductors within a small form factor remains challenging in the standard CMOS process, which is attributed to the cost and technology limitation. However, replacing the inductor with bond wires can improve the form factor compared to conventional SIPC [175]. Increasing ohmic losses with increasing switching frequency, magnetic core losses, hysteresis losses, and eddy current losses associated

12737

TABLE III Advantages and Disadvantages of Various Power Conversion Schemes Used in Implantable Devices

Туре	Advantages	Disadvantages
LDO	 One of the simplest forms of voltage conversion technique Small form-factor High efficiency step-down operation is possible Can produce noise free voltage for noise-sensitive electronics. 	- Voltage up-conversion is not possible - Can produce heat during regulation which can lead to tissue damage - Is not practical for large V_{in} - V_{out}
SCPC	 Mature on-chip integration can yield smaller footprint At low input voltage, cold-start circuitry is not typically required Superior performance compared to LDO when V_{in}-V_{out} is large PFM ensures negligible switching loss at lighter load and low conversion ratio. 	 Voltage conversion ratio is somewhat limited Limited maximum power handling capability May require large form factor May become expensive due to high-density capacitor fabrication Stepped output voltages, fine tuning is difficult
SIPC	 Suited for high-power, high-efficiency applications Wide range of voltage conversion possible without sacrificing form-factor Continuous voltage conversion ratio AC-DC switched inductor enables maximum power extraction from kinetic harvesters 	 Relatively large footprint and EMI susceptibility due to inductor May result in large RMS current Fabrication of on-chip inductors may require larger area May require additional cold-start circuitry, resulting in increasing cost and footprint

with the inductor add to the overall loss profile of a SIPC topology, thereby compromising converter efficiency [176]. At low output powers, moving to DCM over CCM allow the converter to use small-sized inductors leading to improved form factor and increased efficiency, although at the price of higher current ripple and the complex timing of the switch driving signals. DCM operation also facilitates the time-sharing of a single inductor in a multisource and/or multioutput platform, thereby can maintain a comparatively smaller form factor.

As a conclusion to this section, Table III briefly states the advantages and disadvantages of each type of power converters used in implantable devices discussed in this section.

B. Comparison of SCPC and SIPC Based on Key Performance Metrics

In this section, a comparative analysis of SCPC and SIPC suitable for implantable electronics is presented based on their electrical parameters and performances. Since this is a quantitative comparison based on key performance metrics, we reviewed only dc–dc switched-mode converters in order to perform an apple to apple comparison. We compared the peak efficiencies of these two converters with respect to different performance metrics in Fig. 21. Maximum output power, power density, VCR, and input voltage are the selected metrics that are individually plotted against peak efficiency to compare the performance of SIPC and SCPC converters reported in the literature.

Fig. 21(a) illustrates the comparison of peak efficiency versus output power for the two converter topologies. It highlights the widespread use of SCPC in ultralow ($\sim 0.1 \ \mu$ W) to low power ($\sim 1 \ m$ W) implantable loads. At relatively high-power levels (>1 mW), the required number of flying capacitors significantly increases, resulting in compromised form factor and increased switching loss, thereby hindering the feasibility of utilizing SCPC. In contrast, except single-stage ac–dc SIPC (exists at microwatt level), dc–dc SIPC is limited to a relatively high-power levels ranging from $\sim 1 \ m$ W to $\sim 30 \ W$ due to the implementation of discrete inductors. SIPC converters are traditionally configured in buck, boost, and/or buck–boost configuration based on a single inductor that allows voltage up/down conversion without

significant area penalty compared to SCPC. An SIPC topology reported in [20] has achieved a maximum output power close to 30 W, and it substantiates the superior ability of SIPC to operate at higher output power. Fig. 20(b) illustrates the number of existing SCPCs tailing off as power density increases. The decline in existing SCPC topologies at high power density can be attributed to the low capacitive density of on-chip capacitors realized by standard CMOS technology. However, capacitors implemented by superior technology (e.g., trench capacitors [184]) can reach the high end of the power density spectrum with added implementation cost. Unlike SCPC, SIPC does not exhibit any declining trend with an increased power density as evident from Fig. 21(b). In light of the discussion based on Fig. 21(a) and (b), it is worth mentioning that future research efforts in power converter design tailored for a biomedical application must focus on developing SCPC suited for medium-to-high power levels.

The next performance metric is the VCR. The superiority of SIPCs over SCPCs in terms of VCR is distinguishable from Fig. 21(c). SCPCs can be configured to achieve a higher conversion ratio by adding a large number of flying capacitors. However, the resulting VCR improvement comes at a price of larger footprint, increased switching loss, and increased implementation cost. Therefore, most of the SCPCs discussed here are configured to achieve a low to moderate VCR ranging from 1 to 10. In contrast, SIPCs can attain a high VCR without sacrificing a large die area, as the VCR primarily depends on the duty ratio of the circuit. As shown in Fig. 21(c), an SIPC can achieve a VCR as high as \sim 100 [186].

The comparison between the two converter topologies in terms of input voltage is illustrated in Fig. 21(d). The majority of the research on SCPC topology is focused on a narrow input voltage range of 0.2–2 V, whereas SIPC can operate at an input voltage as low as 20 mV [131]. Therefore, SIPC is better suited for applications where the generated voltage by an energy harvester is significantly low. Due to the ability to achieve a high conversion ratio with better efficiency and form factor by changing only the duty ratio, a conventional SIPC can be configured to obtain a high voltage output from a low input voltage ($\sim 20 \text{ mV}$) provided that an appropriate cold-start circuit



Fig. 21. Existing power converter efficiency versus (a) output power, (b) power density, (c) voltage conversion ratio, and (d) input voltage.

is adopted. However, a conventional SCPC can achieve the same feature by adopting a large number of capacitors, but not without the expense of additional cost, area, and switching loss.

Considering the attributes featured in Fig. 21(a)–(d), we can establish a selection process of power converters suitable for the intended biomedical application with a specific voltage and power requirement as there is no "*one-size-fits-all*" solution. For instance, implants that require a high-efficiency power conversion at high output power may adopt SIPC topology. For implants requiring ultralow to low output power can use both SCPC and SIPC. That being said, conventional SIPC requires



Fig. 22. Oscillator-based cold-start circuit. (a) Conventional CMOS inverterbased oscillator circuit [190]. (b) ILRO circuit [130] and transformer oscillation mode boost converter where low-voltage start-up is achieved by the built-in mutual inductance of the transformer [132].

a larger footprint in comparison to an SCPC. Hence, SCPC would be the appropriate choice of power conversion in in-body implants that must maintain a small form factor.

VII. START-UP ISSUES

Many of the energy harvesters discussed so far suffer from the problem of "*cold-start*." This situation occurs when the harvested energy from ambient sources, such as TEGs, MFCs, and solar PV energy sources, yields converted voltage levels that are insufficient to provide ancillary power for the gate drivers and other auxiliary circuits/functional blocks [23], [28].

One approach to achieving start-up is to adopt oscillator-based circuits, where an oscillator is incorporated, usually followed by a VM circuit or a charge pump. The oscillator uses the low input voltage of the harvester to generate a clock signal that is boosted beyond the supply voltage of the VM circuit to facilitate the start-up. The easiest oscillator structure is the conventional CMOS inverter-based oscillator or ring oscillator [133], [189], [190], as shown in Fig. 22(a). The threshold voltage of MOSFETs limits the minimum input voltage of the oscillator, which makes it unattractive for a low-voltage start-up circuit. However, it is possible to reduce the start-up voltage by adopting some postprocessing threshold voltage tuning, which imposes additional processing costs. LC tank oscillator and inductive load ring oscillator (ILRO) based cold-start systems can overcome the above-mentioned problem related to MOSFET threshold voltage and can achieve as low as half of the start-up voltage of a classical inverter-based ring oscillator [130], [134], [143]. Fig. 22(b) shows an ILRO-based start-up architecture that obtains oscillation by using two inductors configured in mutual feedback with the help of a cross-coupled transistor pair. Like the ILRO-based start-up scheme, mutual feedback can be realized by the built-in

mutual inductance of a transformer due to the common magnetic core sharing between inductors. Such a transformer-based self-staring boost converter for a TEG harvester is shown in a simplified manner in Fig. 22(c) [132]. This cold-start circuit can achieve a minimum self-start-up voltage of 21 mV through a positive feedback loop formed by a native ZVT MOSFET (M_{TOM}) together with the transformer winding pair, which is triggered by thermal noise. This transformer-based cold-start solution requires a very high turn ratio to achieve direct multiplication of the harvester's low input voltage, which increases the OFF-state leakage current of the native MOSFET. To overcome this issue, Teh and Mok [140] introduced a modified boost converter to initiate the start-up that utilizes a single switch-based unity turns ratio transformer, such as a high inductance pulse transformer. However, both inductor- and transformer-based start-up schemes require additional off-chip magnetic components that limit device miniaturization. Recently, Garcha et al. proposed a fully integrated start-up solution using a Meissner Oscillator with an on-chip transformer that could achieve start-up voltage as low as 25 mV [191].

Rather than using an oscillator circuit, some external sources such as a battery or supercapacitor [77], [131], [133], [141] and auxiliary RF energy source [133], [136] were used to carry out one-time precharging of the load/output capacitor [131], [133], [136], [141]. The size of the output/load capacitor is set for comparatively high-power requirements of the load rather than the control circuit. This implies that a comparatively large amount of power is needed to charge it during the start-up period resulting in poor efficiency of the total power conversion system. This issue was addressed by Goeppert and Manoli [135] carrying out one-time precharging on a separate storage capacitor that is much smaller than the output capacitor and is independent of the implant load requirements. Nevertheless, the converter output takes over the power delivery to the control circuit and thus affects the PCE again. To overcome this problem, a separate dedicated output to power the control circuit rather than powering the implant load during the start-up as well as the regular operation is proposed in [61] and [138]. Fig. 14 can be referred for realizing such a start-up scheme that was able to help the boost converter operate with input voltages as low as 15 mV.

Unlike storing sufficient energy by stepping up the source voltage for the cold-start, direct triggering of the active components was reported that include the additional/separate external sources such as battery [113] and mechanical switch [139]. For instance, Ramadass and Chandrakasan designed a mechanically assisted start-up method for a TEG-supplied boost converter, which can achieve start-up at an input voltage as low as 35 mV, and the circuit diagram of the energy harvester is already shown in Fig. 12. However, separate external sources are comparatively expensive and require off-chip components that limit their applications.

The design of the cold-start circuit is application dependent. For example, while designing a cold-start circuit for ultralowvoltage energy sources such as TEG, MFCs, or solar, the design considerations need to be focused on minimizing start-up voltage and the time needed to initialize the start-up, even at the cost of lower efficiency. In contrast, for other energy harvesting applications, the cold-start is considered an inefficient process and is often excluded from published efficiency values due to the event rarely occurring, possibly as little as once over the lifetime of the system [28].

VIII. EXISTING RESEARCH GAP

For any given application, finding the most suitable power converter topology seems to be the biggest hurdle because there is no universal architecture that performs the best for all applications. Many uncertainties exist, and the key factors behind the selection of any converter are type of implant, type of energy source, voltage and current ratings, nature of power-pulsed or dc, allowable volume, EMI and magnetic resonance imaging (MRI) related issues, and so on. With IPT or kinetic energy harvesting, ac-dc conversion is needed. Most of the high-performance ac-dc or dc-dc converters employ inductors-either on-chip or off-chip, and they suffer from EMI issues and start-up problems. With large implanted inductors, patients cannot undergo MRI. LDO-based systems are effective as long as the voltage drop across the converter is small, and they may suffer from heat dissipation within the body once the voltage drop exceeds a limit. Therefore, we can identify the technology gaps in the following way. First, there is a clear disjoint between the biomedical community and the power electronics community-the researchers designing sensors and implanted circuits do not have a clear understanding of the power processing architectures. In many cases, the proposed solution is not the most optimum. Second, because there is no universal solution, researchers are thinking of designing switching power converters without inductive elements. In this way, the VCR of the converters can be controlled very precisely as opposed to SC-based converters. During the last ten years, the coauthors of this article have proposed a MEMS resonator-based power conversion architecture that can achieve high VCR without using inductors [192]-[194]. This type of "system-on-chip" solutions are still in their infancy, and extensive research is needed to come up with a universal solution that could be implanted without any EMI or MRI issues. Because resonators are electromechanical energy storage devices, power converter circuits based around them do not generate EMIs, which could be a key element to decide the optimum power converter topology for any given implantable application.

IX. CONCLUSION

This review article summarizes a plethora of state-of-theart power conditioning circuits that can establish end-to-end power conditioning interfaces between in-body energy harvesting sources and target implant applications. Rectification, voltage regulation, and auxiliary circuit for maximum power transfer and cold-start constitute the basic end-to-end power conditioning interface for an implantable system. The uniqueness of each biomedical application creates a unique set of voltage and power requirements. Hence, elaborate knowledge on the strengths and limitations of each power conversion architecture and its working mechanism is of paramount importance. This knowledge can help the researchers to determine the power conversion block that is the best fit for a specific set of implant load and energy harvesting/power delivery schemes.

With the recent advancements in power electronics technology, researchers are moving away from simple passive topologies toward advanced self-powered switching regulators. The active/synchronous full-wave rectifiers are more efficient than their passive counterparts; however, if a $V_{\rm th}$ mitigation technique and/or low-voltage threshold process is adopted, then passive rectifiers could possibly offer superior performance at lower power levels of implant application. VM are basically rectifying technologies with relatively poor voltage regulation ability. In designs where direct ac-dc conversion with regulated output voltage is required, single-stage direct ac-dc switched-mode converters can become a feasible solution. These single-stage converters are able to provide optimal impedance match to the low-power kinetic energy harvester in order to harvest maximum power. We have also reviewed two-stage power conversion topologies for implant applications, where the rectification stage is separate from the voltage conditioning and/or regulation stage. LDO is found to often offer a better choice compared to switched-mode converters for the circuits that require less than a few hundred milliwatts, provided that $V_{\rm in}-V_{\rm out}$ is small, and only voltage down-conversion is needed. Both types of dc-dc switched-mode converters, namely SCs and switched inductors, enable efficient up and down-conversion. Using PFM instead of PWM, SC converters offer better performance at comparatively ultralow (~0.1 μ W) to low power (~1 mW) implantable loads. Compared to SC converters, switched-inductor converters are generally configured for higher output power IMDs (>1 mW) without significant area penalty. However, SC converters offering similar output power in the range of switched-inductor converters are appearing, whereas switched-inductor converters are less suitable for ultralow output powers. Nevertheless, EMI generated from the inductors used in inductor-based converters may pose a substantial health risk to patients hosting the associated IMDs. A micro-electromechanical system (MEMS) resonator-based converter can be an intriguing alternative to such inductor-based converters to mitigate EMI-related health hazards.

REFERENCES

- [1] A. B. Amar, A. B. Kouki, and H. Cao, "Power approaches for implantable medical devices," *Sensors*, vol. 15, no. 11, pp. 28889–28914, Nov. 2015, doi: 10.3390/s151128889.
- [2] R. J. M. Vullers, R. van Schaijk, I. Doms, C. Van Hoof, and R. Mertens, "Micropower energy harvesting," *Solid-State Electron.*, vol. 53, no. 7, pp. 684–693, Jul. 2009, doi: 10.1016/j.sse.2008.12.011.
- [3] J. M. Khalifeh *et al.*, "Electrical stimulation and bone healing: A review of current technology and clinical applications," *IEEE Rev. Biomed. Eng.*, vol. 11, pp. 217–232, 2018, doi: 10.1109/RBME.2018.2799189.
- [4] R. Schnell, C. Graeff, A. Krebs, H. Oertel, and C.-C. Glüer, "Changes in cross-sectional area of spinal canal and vertebral body under 2 years of teriparatide treatment: Results from the EUROFORS study," *Calcified Tissue Int.*, vol. 87, no. 2, pp. 130–136, Aug. 2010, doi: 10.1007/s00223-010-9386-8.
- [5] C.-Y. Tsui, X. Li, and W.-H. Ki, "Energy harvesting and power delivery for implantable medical devices," *Found. Trends Electron. Des. Autom.*, vol. 7, no. 3, pp. 179–246, Aug. 2013, doi: 10.1561/100000029.
- [6] S. Lee, C. Hsieh, and C. Yang, "Wireless front-end with power management for an implantable cardiac microstimulator," *IEEE Trans. Biomed. Circuits Syst.*, vol. 6, no. 1, pp. 28–38, Feb. 2012.

- [7] L. S. Y. Wong, S. Hossain, A. Ta, J. Edvinsson, D. H. Rivas, and H. Naas, "A very low-power CMOS mixed-signal IC for implantable pacemaker applications," *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2446–2456, Dec. 2004.
- [8] J. Charthad *et al.*, "An ultrasonically powered implantable device for targeted drug delivery," in *Proc. 38th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2016, pp. 541–544, doi: 10.1109/EMBC.2016.7590759.
- [9] B. Yan et al., "Battery-free implantable insulin micropump operating at transcutaneously radio frequency-transmittable power," Med. Devices Sens., vol. 2, no. 5/6, 2019, Art. no. e10055, doi: 10.1002/mds3.10055.
- [10] J. Yoo *et al.*, "A 5.2 mW self-configured wearable body sensor network controller and a 12 μW 54.9% efficiency wirelessly powered sensor for continuous health monitoring system," in *Proc. IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, San Francisco, CA, USA, Feb. 2009, pp. 290–291, doi: 10.1109/ISSCC.2009.4977422.
- [11] S. Lee, L. Yan, T. Roh, S. Hong, and H. Yoo, "A 75 μW real-time scalable body area network controller and a 25 μW ExG sensor IC for compact sleep monitoring applications," *IEEE J. Solid-State Circuits*, vol. 47, no. 1, pp. 323–334, Jan. 2012.
- [12] M. Yip, R. Jin, H. H. Nakajima, K. M. Stankovic, and A. P. Chandrakasan, "A fully-implantable cochlear implant SoC with piezoelectric middle-ear sensor and arbitrary waveform neural stimulation," *IEEE J. Solid-State Circuits*, vol. 50, no. 1, pp. 214–229, Jan. 2015.
- [13] M. W. Baker, "A low-power cochlear implant system," Ph.D. dissertation, Dept. Elect. Eng. Comput. Sci., Massachusetts Inst. Technol., Cambridge, MA, USA, 2007.
- [14] S. Greenberg, W. A. Ainsworth, and R. R. Fay, Eds., Speech Processing in the Auditory System. New York, NY, USA: Springer-Verlag, 2004.
- [15] M. H. Maghami, A. M. Sodagar, and M. Sawan, "Versatile stimulation back-end with programmable exponential current pulse shapes for a retinal visual prosthesis," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 24, no. 11, pp. 1243–1253, Nov. 2016.
- [16] S. Ethier and M. Sawan, "Exponential current pulse generation for efficient very high-impedance multisite stimulation," *IEEE Trans. Biomed. Circuits Syst.*, vol. 5, no. 1, pp. 30–38, Feb. 2011.
- [17] Y. Lo, K. Chen, P. Gad, and W. Liu, "A fully-integrated high-compliance voltage SoC for epi-retinal and neural prostheses," *IEEE Trans. Biomed. Circuits Syst.*, vol. 7, no. 6, pp. 761–772, Dec. 2013.
- [18] Y. Lo *et al.*, "A fully integrated wireless SoC for motor function recovery after spinal cord injury," *IEEE Trans. Biomed. Circuits Syst.*, vol. 11, no. 3, pp. 497–509, Jun. 2017.
- [19] C. Hsu, S. Tseng, Y. Hsieh, and C. Wang, "One-time-implantable spinal cord stimulation system prototype," *IEEE Trans. Biomed. Circuits Syst.*, vol. 5, no. 5, pp. 490–498, Oct. 2011.
- [20] O. Knecht and J. W. Kolar, "Performance evaluation of seriescompensated IPT systems for transcutaneous energy transfer," *IEEE Trans. Power Electron.*, vol. 34, no. 1, pp. 438–451, Jan. 2019.
- [21] O. Knecht, R. Bosshard, and J. W. Kolar, "High-efficiency transcutaneous energy transfer for implantable mechanical heart support systems," *IEEE Trans. Power Electron.*, vol. 30, no. 11, pp. 6221–6236, Nov. 2015.
- [22] P. D. Mitcheson and T. T. Toh, "Power management electronics," in *Energy Harvesting for Autonomous Systems*. Norwood, MA, USA: Artech House, 2010.
- [23] G. D. Szarka, B. H. Stark, and S. G. Burrow, "Review of power conditioning for kinetic energy harvesting systems," *IEEE Trans. Power Electron.*, vol. 27, no. 2, pp. 803–815, Feb. 2012.
- [24] S. Cheng, Y. Jin, Y. Rao, and D. P. Arnold, "An active voltage doubling AC/DC converter for low-voltage energy harvesting applications," *IEEE Trans. Power Electron.*, vol. 26, no. 8, pp. 2258–2265, Aug. 2011.
- [25] G. A. Rincon-Mora and P. E. Allen, "A low-voltage, low quiescent current, low drop-out regulator," *IEEE J. Solid-State Circuits*, vol. 33, no. 1, pp. 36–44, Jan. 1998.
- [26] G. Villar-Piqué, H. J. Bergveld, and E. Alarcón, "Survey and benchmark of fully integrated switching power converters: Switched-capacitor versus inductive approach," *IEEE Trans. Power Electron.*, vol. 28, no. 9, pp. 4156–4167, Sep. 2013.
- [27] A. P. Chandrakasan, N. Verma, and D. C. Daly, "Ultralow-power electronics for biomedical applications," *Annu. Rev. Biomed. Eng.*, vol. 10, pp. 247–274, 2008, doi: 10.1146/annurev.bioeng.10.061807.160547.
- [28] D. Newell and M. Duffy, "Review of power conversion and energy management for low-power, low-voltage energy harvesting powered wireless sensors," *IEEE Trans. Power Electron.*, vol. 34, no. 10, pp. 9794–9805, Oct. 2019.
- [29] IEEE Standard for Safety Levels With Respect to Human Exposure to Radio Frequency Electromagnetic Fields, 3 kHz To 300 GHz, IEEE Standard C951-2005, Rev. IEEE Standard C951-1991, pp. 1–238, Apr. 2006, doi: 10.1109/IEEESTD.2006.99501.

- [30] D. Flynn, "Challenges for power supplies in medical equipment: Ensuring patient and operator safety," *IEEE Power Electron. Mag.*, vol. 2, no. 2, pp. 32–37, Jun. 2015.
- [31] P. Li and R. Bashirullah, "A wireless power interface for rechargeable battery operated medical implants," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 54, no. 10, pp. 912–916, Oct. 2007.
- [32] Y. Hwang, C. Lei, Y. Yang, J. Chen, and C. Yu, "A 13.56-MHz low-voltage and low-control-loss RF-DC rectifier utilizing a reducing reverse loss technique," *IEEE Trans. Power Electron.*, vol. 29, no. 12, pp. 6544–6554, Dec. 2014.
- [33] H. Uluşan, S. Chamanian, Ö. Zorlu, A. Muhtaroğlu, and H. Külah, "Neural stimulation interface with ultra-low power signal conditioning circuit for fully-implantable cochlear implants," in *Proc. IEEE Biomed. Circuits Syst. Conf.*, Oct. 2017, pp. 1–4, doi: 10.1109/BIOCAS.2017.8325060.
- [34] M. Usami, A. Sato, K. Sameshima, K. Watanabe, H. Yoshigi, and R. Imura, "Powder LSI: An ultra small RF identification chip for individual recognition applications," in *Proc. IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, Feb. 2003, vol. 1, pp. 398–501, doi: 10.1109/ISSCC.2003.1234354.
- [35] T. T. Le, J. Han, A. von Jouanne, K. Mayaram, and T. S. Fiez, "Piezoelectric micro-power generation interface circuits," *IEEE J. Solid-State Circuits*, vol. 41, no. 6, pp. 1411–1420, Jun. 2006.
- [36] L. Dongsheng, L. Huan, Z. Xuecheng, G. Liang, Y. Ke, and L. Zilong, "A high sensitivity analog Front-end circuit for semi-passive HF RFID tag applied to implantable devices," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 62, no. 8, pp. 1991–2002, Aug. 2015.
- [37] A. Facen and A. Boni, "Power supply generation in CMOS passive UHF RFID tags," in *Proc. Ph.D. Res. Microelectron. Electron.*, Jun. 2006, pp. 33–36, doi: 10.1109/RME.2006.1689889.
- [38] J. H. Cheong *et al.*, "An inductively powered implantable blood flow sensor microsystem for vascular grafts," *IEEE Trans. Biomed. Eng.*, vol. 59, no. 9, pp. 2466–2475, Sep. 2012.
- [39] Z. Wang and S. Mirabbasi, "A low-voltage CMOS rectifier with on-chip matching network and a magnetic field focused antenna for wirelessly powered medical implants," *IEEE Trans. Biomed. Circuits Syst.*, vol. 13, no. 3, pp. 554–565, Jun. 2019.
- [40] M. H. Ouda, W. Khalil, and K. N. Salama, "Self-biased differential rectifier with enhanced dynamic range for wireless powering," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 64, no. 5, pp. 515–519, May 2017.
- [41] T. Umeda, H. Yoshida, S. Sekine, Y. Fujita, T. Suzuki, and S. Otaka, "A 950-MHz rectifier circuit for sensor network tags with 10-m distance," *IEEE J. Solid-State Circuits*, vol. 41, no. 1, pp. 35–41, Jan. 2006.
- [42] M. A. Al-Absi and S. R. Al-Batati, "Hybrid internal vth cancellation rectifiers for RF energy harvesting," *IEEE Access*, vol. 8, pp. 51976–51980, 2020.
- [43] K. Kotani, and T. Ito, "High efficiency CMOS rectifier circuit with self-Vth-cancellation and power regulation functions for UHF RFIDs," in *Proc. IEEE Asian Solid-State Circuits Conf.*, Nov. 2007, pp. 119–122, doi: 10.1109/ASSCC.2007.4425746.
- [44] D.-S. Liu, F.-B. Li, X.-C. Zou, Y. Liu, X.-M. Hui, and X.-F. Tao, "New analysis and design of a RF rectifier for RFID and implantable devices," *Sensors*, vol. 11, no. 7, pp. 6494–6508, Jun. 2011, doi: 10.3390/s110706494.
- [45] M. Ghovanloo and K. Najafi, "Fully integrated wideband high-current rectifiers for inductively powered devices," *IEEE J. Solid-State Circuits*, vol. 39, no. 11, pp. 1976–1984, Nov. 2004.
- [46] S. Lee, J. Hong, C. Hsieh, M. Liang, and J. Kung, "A low-power 13.56 MHz RF front-end circuit for implantable biomedical devices," *IEEE Trans. Biomed. Circuits Syst.*, vol. 7, no. 3, pp. 256–265, Jun. 2013.
- [47] C. Peters, F. Henrici, M. Ortmanns, and Y. Manoli, "High-bandwidth floating gate CMOS rectifiers with reduced voltage drop," in *Proc. IEEE Int. Symp. Circuits Syst.*, May 2008, pp. 2598–2601, doi: 10.1109/IS-CAS.2008.4541988.
- [48] Y. Berg, D. T. Wisland, and T. S. Lande, "Ultra low-voltage/low-power digital floating-gate circuits," *IEEE Trans. Circuits Syst. II, Analog Digit. Signal Process.*, vol. 46, no. 7, pp. 930–936, Jul. 1999.
- [49] S. S. Hashemi, M. Sawan, and Y. Savaria, "A high-efficiency low-voltage CMOS rectifier for harvesting energy in implantable devices," *IEEE Trans. Biomed. Circuits Syst.*, vol. 6, no. 4, pp. 326–335, Aug. 2012.
- [50] S. R. Khan and G. Choi, "High-efficiency CMOS rectifier with minimized leakage and threshold cancellation features for low power bio-implants," *Microelectron. J.*, vol. 66, pp. 67–75, Aug. 2017, doi: 10.1016/j.mejo.2017.06.002.

- [51] S. Hashemi, M. Sawan, and Y. Savaria, "A novel low-drop CMOS active rectifier for RF-powered devices: Experimental results," *Microelectron. J.*, vol. 40, no. 11, pp. 1547–1554, Nov. 2009, doi: 10.1016/j.mejo.2009.02.007.
- [52] Z. Hameed and K. Moez, "A 3.2 v -15 dBm adaptive thresholdvoltage compensated RF energy harvester in 130 nm CMOS," *IEEE Trans. Circuits Syst. Regul. Papers*, vol. 62, no. 4, pp. 948–956, Apr. 2015.
- [53] D. Rozgić et al., "A 0.338 cm³, artifact-free, 64-Contact neuromodulation platform for simultaneous stimulation and sensing," *IEEE Trans. Biomed. Circuits Syst.*, vol. 13, no. 1, pp. 38–55, Feb. 2019.
- [54] H. Cha, W. Park, and M. Je, "A CMOS rectifier with a cross-coupled latched comparator for wireless power transfer in biomedical applications," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 59, no. 7, pp. 409–413, Jul. 2012.
- [55] H. Cheng, C. A. Gong, and S. Kao, "A 13.56 MHz CMOS high-efficiency active rectifier with dynamically controllable comparator for biomedical wireless power transfer systems," *IEEE Access*, vol. 6, pp. 49979–49989, 2018.
- [56] Y. Lo, K. Chen, P. Gad, and W. Liu, "An on-chip multi-voltage power converter with leakage current prevention using 0.18μm high-voltage CMOS process," *IEEE Trans. Biomed. Circuits Syst.*, vol. 10, no. 1, pp. 163–174, Feb. 2016.
- [57] Y. Lu and W. Ki, "A 13.56 MHz CMOS active rectifier with switchedoffset and compensated biasing for biomedical wireless power transfer systems," *IEEE Trans. Biomed. Circuits Syst.*, vol. 8, no. 3, pp. 334–344, Jun. 2014.
- [58] S. Guo and H. Lee, "An efficiency-enhanced integrated CMOS rectifier with comparator-controlled switches for transcutaneous powered implants," in *Proc. IEEE Custom Integr. Circuits Conf.*, Sep. 2007, pp. 385–388, doi: 10.1109/CICC.2007.4405758.
- [59] T. Lehmann and Y. Moghe, "On-chip active power rectifiers for biomedical applications," in *Proc. IEEE Int. Symp. Circuits Syst.*, May 2005, vol. 1, pp. 732–735, doi: 10.1109/ISCAS.2005.1464692.
- [60] W. Chen et al., "A fully integrated 8-Channel closed-loop neuralprosthetic CMOS SoC for real-time epileptic seizure control," *IEEE J. Solid-State Circuits*, vol. 49, no. 1, pp. 232–247, Jan. 2014.
- [61] J. Charthad *et al.*, "A mm-sized wireless implantable device for electrical stimulation of peripheral nerves," *IEEE Trans. Biomed. Circuits Syst.*, vol. 12, no. 2, pp. 257–270, Apr. 2018.
- [62] Y. Lam, W. Ki, and C. Tsui, "Integrated low-loss CMOS active rectifier for wirelessly powered devices," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 53, no. 12, pp. 1378–1382, Dec. 2006.
- [63] H. Uluşan, Ö. Zorlu, A. Muhtaroğlu, and H. Külah, "Highly integrated 3 v supply electronics for electromagnetic energy harvesters with minimum 0.4 v_{peak} input," *IEEE Trans. Ind. Electron.*, vol. 64, no. 7, pp. 5460–5467, Jul. 2017.
- [64] J. Sankman and D. Ma, "A 12-μW to 1.1-mW AIM piezoelectric energy harvester for time-varying vibrations with 450-nA "bm Q," *IEEE Trans. Power Electron.*, vol. 30, no. 2, pp. 632–643, Feb. 2015.
- [65] S. Guo and H. Lee, "An efficiency-enhanced CMOS rectifier with unbalanced-biased comparators for transcutaneous-powered highcurrent implants," *IEEE J. Solid-State Circuits*, vol. 44, no. 6, pp. 1796–1804, Jun. 2009.
- [66] D. Pivonka, A. Yakovlev, A. S. Y. Poon, and T. Meng, "A mm-sized wirelessly powered and remotely controlled locomotive implant," *IEEE Trans. Biomed. Circuits Syst.*, vol. 6, no. 6, pp. 523–532, Dec. 2012.
- [67] H. Miura, S. Arai, Y. Kakubari, F. Sato, H. Matsuki, and T. Sato, "Improvement of the transcutaneous energy transmission system utilizing ferrite cored coils for artificial hearts," *IEEE Trans. Magn.*, vol. 42, no. 10, pp. 3578–3580, Oct. 2006.
- [68] H. Lee and M. Ghovanloo, "An adaptive reconfigurable active voltage doubler/rectifier for extended-range inductive power transmission," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 59, no. 8, pp. 481–485, Aug. 2012.
- [69] Z. J. Chew and M. Zhu, "Adaptive self-configurable rectifier for extended operating range of piezoelectric energy harvesting," *IEEE Trans. Ind. Electron.*, vol. 67, no. 4, pp. 3267–3276, Apr. 2020.
- [70] A. Tabesh and L. G. Frechette, "A low-power stand-alone adaptive circuit for harvesting energy from a piezoelectric micropower generator," *IEEE Trans. Ind. Electron.*, vol. 57, no. 3, pp. 840–849, Mar. 2010.
- [71] C. M. Tan et al., "A possible reality on battery-less low-power portable electronics," in Proc. 5th IEEE Conf. Ind. Electron. Appl., Jun. 2010, pp. 630–633, doi: 10.1109/ICIEA.2010.5517024.

- [72] R. Torah, P. Glynne-Jones, M. Tudor, T. O'Donnell, S. Roy, and S. Beeby, "Self-powered autonomous wireless sensor node using vibration energy harvesting," *Meas. Sci. Technol.*, vol. 19, no. 12, Oct. 2008, Art. no. 125202, doi: 10.1088/0957-0233/19/12/125202.
- [73] J. Charthad, M. J. Weber, T. C. Chang, and A. Arbabian, "A mm-sized implantable medical device (IMD) with ultrasonic power transfer and a hybrid bi-directional data link," *IEEE J. Solid-State Circuits*, vol. 50, no. 8, pp. 1741–1753, Aug. 2015.
- [74] H.-M. Lee and M. Ghovanloo, "A high frequency active voltage doubler in standard CMOS using offset-controlled comparators for inductive power transmission," *IEEE Trans. Biomed. Circuits Syst.*, vol. 7, no. 3, pp. 213–224, Jun. 2013.
- [75] F. Mounaim and M. Sawan, "Integrated high-voltage inductive power and data-recovery front end dedicated to implantable devices," *IEEE Trans. Biomed. Circuits Syst.*, vol. 5, no. 3, pp. 283–291, Jun. 2011.
- [76] C. K. Tse, S. C. Wong, and M. H. L. Chow, "On lossless switchedcapacitor power converters," *IEEE Trans. Power Electron.*, vol. 10, no. 3, pp. 286–291, May 1995.
- [77] L. Wu, X. Do, S. Lee, and D. S. Ha, "A self-powered and optimal SSHI circuit integrated with an active rectifier for piezoelectric energy harvesting," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 64, no. 3, pp. 537–549, Mar. 2017.
- [78] S. Dwari and L. Parsa, "Efficient low voltage direct AC/DC converters for self-powered wireless sensor nodes and mobile electronics," in *Proc. IEEE 30th Int. Telecommun. Energy Conf.*, Sep. 2008, pp. 1–7, doi: 10.1109/INTLEC.2008.4664097.
- [79] S. Dwari, R. Dayal, and L. Parsa, "A novel direct AC/DC converter for efficient low voltage energy harvesting," in *Proc.* 34th Annu. Conf. IEEE Ind. Electron., Nov. 2008, pp. 484–488, doi: 10.1109/IECON.2008.4758001.
- [80] P. D. Mitcheson, T. C. Green, and E. M. Yeatman, "Power processing circuits for electromagnetic, electrostatic and piezoelectric inertial energy scavengers," *Microsyst. Technol.*, vol. 13, no. 11/12, pp. 1629–1635, Jul. 2007, doi: 10.1007/s00542-006-0339-0.
- [81] R. Dayal, S. Dwari, and L. Parsa, "Design and implementation of a direct AC–DC boost converter for low-voltage energy harvesting," *IEEE Trans. Ind. Electron.*, vol. 58, no. 6, pp. 2387–2396, Jun. 2011.
- [82] S. Dwari and L. Parsa, "An efficient AC–DC step-up converter for lowvoltage energy harvesting," *IEEE Trans. Power Electron.*, vol. 25, no. 8, pp. 2188–2199, Sep. 2010.
- [83] R. Bolt, M. Magno, T. Burger, A. Romani, and L. Benini, "Kinetic AC/DC converter for electromagnetic energy harvesting in autonomous wearable devices," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 64, no. 12, pp. 1422–1426, Dec. 2017.
- [84] B. H. Stark, P. D. Mitcheson, P. Miao, T. C. Green, E. M. Yeatman, and A. S. Holmes, "Converter circuit design, semiconductor device selection and analysis of parasitics for micropower electrostatic generators," *IEEE Trans. Power Electron.*, vol. 21, no. 1, pp. 27–37, Jan. 2006.
- [85] D. Guyomar and M. Lallart, "Recent progress in piezoelectric conversion and energy harvesting using nonlinear electronic interfaces and issues in small scale implementation," *Micromachines*, vol. 2, no. 2, pp. 274–294, Jun. 2011, doi: 10.3390/mi2020274.
- [86] J. Liang and H. S.-H. Chung, "Best voltage bias-flipping strategy towards maximum piezoelectric power generation," J. Phys. Conf. Ser., vol. 476, Dec. 2013, Art. no. 012025, doi: 10.1088/1742-6596/476/1/012025.
- [87] P. C.-P Chao, "Energy harvesting electronics for vibratory devices in self-powered sensors," *IEEE Sens. J.*, vol. 11, no. 12, pp. 3106–3121, Dec. 2011.
- [88] D. Guyomar, A. Badel, E. Lefeuvre, and C. Richard, "Toward energy harvesting using active materials and conversion improvement by nonlinear processing," *IEEE Trans. Ultrason. Ferroelectr. Freq. Control*, vol. 52, no. 4, pp. 584–595, Apr. 2005.
- [89] K. Makihara, J. Onoda, and T. Miyakawa, "Low energy dissipation electric circuit for energy harvesting," *Smart Mater. Struct.*, vol. 15, no. 5, pp. 1493–1498, Sep. 2006, doi: 10.1088/0964-1726/15/5/039.
- [90] L. Garbuio, M. Lallart, D. Guyomar, C. Richard, and D. Audigier, "Mechanical energy harvester with ultralow threshold rectification based on SSHI nonlinear technique," *IEEE Trans. Ind. Electron.*, vol. 56, no. 4, pp. 1048–1056, Apr. 2009.
- [91] M. Lallart, C. Richard, L. Garbuio, L. Petit, and D. Guyomar, "High efficiency, wide load bandwidth piezoelectric energy scavenging by a hybrid nonlinear approach," *Sens. Actuators Phys.*, vol. 165, no. 2, pp. 294–302, Feb. 2011.
- [92] M. Lallart and D. Guyomar, "Piezoelectric conversion and energy harvesting enhancement by initial energy injection," *Appl. Phys. Lett.*, vol. 97, no. 1, Jul. 2010, Art. no. 014104, doi: 10.1063/1.3462304.

- [93] A. Badel, D. Guyomar, E. Lefeuvre, and C. Richard, "Efficiency enhancement of a piezoelectric energy harvesting device in pulsed operation by synchronous charge inversion," *J. Intell. Mater. Syst. Struct.*, vol. 16, no. 10, pp. 889–901, Oct. 2005, doi: 10.1177/1045389X05053150.
- [94] L. Wu and D. S. Ha, "A self-powered piezoelectric energy harvesting circuit with an optimal flipping time SSHI and maximum power point tracking," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 66, no. 10, pp. 1758–1762, Oct. 2019.
- [95] D. A. Sanchez, J. Leicht, E. Jodka, E. Fazel, and Y. Manoli, "21.2 A 4 μWto-1 mW parallel-SSHI rectifier for piezoelectric energy harvesting of periodic and shock excitations with inductor sharing, cold start-up and up to 681% power extraction improvement," in *Proc. IEEE Int. Solid-State Circuits Conf.*, Jan. 2016, pp. 366–367, doi: 10.1109/ISSCC.2016.7418059.
- [96] Y. K. Ramadass and A. P. Chandrakasan, "An efficient piezoelectric energy harvesting interface circuit using a bias-flip rectifier and shared inductor," *IEEE J. Solid-State Circuits*, vol. 45, no. 1, pp. 189–204, Jan. 2010.
- [97] E. E. Aktakka and K. Najafi, "A micro inertial energy harvesting platform with self-supplied power management circuit for autonomous wireless sensor nodes," *IEEE J. Solid-State Circuits*, vol. 49, no. 9, pp. 2017–2029, Sep. 2014.
- [98] S. Lu and F. Boussaid, "A highly efficient P-SSHI rectifier for piezoelectric energy harvesting," *IEEE Trans. Power Electron.*, vol. 30, no. 10, pp. 5364–5369, Oct. 2015.
- [99] M. Dini, A. Romani, M. Filippi, and M. Tartagni, "A nanopower synchronous charge extractor IC for low-voltage piezoelectric energy harvesting with residual charge inversion," *IEEE Trans. Power Electron.*, vol. 31, no. 2, pp. 1263–1274, Feb. 2016.
- [100] E. Lefeuvre, A. Badel, C. Richard, and D. Guyomar, "Piezoelectric energy harvesting device optimization by synchronous electric charge extraction," *J. Intell. Mater. Syst. Struct.*, vol. 16, no. 10, pp. 865–876, Oct. 2005, doi: 10.1177/1045389X05056859.
- [101] Y. Wu, A. Badel, F. Formosa, W. Liu, and A. E. Agbossou, "Piezoelectric vibration energy harvesting by optimized synchronous electric charge extraction," *J. Intell. Mater. Syst. Struct.*, vol. 24, no. 12, pp. 1445–1458, Aug. 2013, doi: 10.1177/1045389X12470307.
- [102] W. Liu, A. Badel, F. Formosa, Q. Zhu, C. Zhao, and G.-D. Hu, "A comprehensive analysis and modeling of the self-powered synchronous switching harvesting circuit with electronic breakers," *IEEE Trans. Ind. Electron.*, vol. 65, no. 5, pp. 3899–3909, May 2018.
- [103] A. Shareef, W. L. Goh, S. Narasimalu, and Y. Gao, "A rectifier-less AC–DC interface circuit for ambient energy harvesting from low-voltage piezoelectric transducer array," *IEEE Trans. Power Electron.*, vol. 34, no. 2, pp. 1446–1457, Feb. 2019.
- [104] E. Lefeuvre, A. Badel, A. Brenes, S. Seok, M. Woytasik, and C.-S. Yoo, "Analysis of piezoelectric energy harvesting system with tunable SECE interface," *Smart Mater. Struct.*, vol. 26, no. 3, Feb. 2017, Art. no. 035065, doi: 10.1088/1361-665X/aa5e92.
- [105] H. Xia, Y. Xia, Y. Ye, L. Qian, G. Shi, and R. Chen, "Analysis and simulation of synchronous electric charge partial extraction technique for efficient piezoelectric energy harvesting," *IEEE Sens. J.*, vol. 18, no. 15, pp. 6235–6244, Aug. 2018.
- [106] A. Morel, P. Gasnier, Y. Wanderoild, G. Pillonnet, and A. Badel, "Short circuit synchronous electric charge extraction (SC-SECE) strategy for wideband vibration energy harvesting," in *Proc.IEEE Int. Symp. Circuits Syst.*, May 2018, pp. 1–5, doi: 10.1109/ISCAS.2018.8351559.
- [107] P. Gasnier et al., "An autonomous piezoelectric energy harvesting IC based on a synchronous multi-shot technique," *IEEE J. Solid-State Cir*cuits, vol. 49, no. 7, pp. 1561–1570, Jul. 2014.
- [108] T. Hehn *et al.*, "A fully autonomous integrated interface circuit for piezoelectric harvesters," *IEEE J. Solid-State Circuits*, vol. 47, no. 9, pp. 2185–2198, Sep. 2012.
- [109] A. Morel et al., "A shock-optimized SECE integrated circuit," IEEE J. Solid-State Circuits, vol. 53, no. 12, pp. 3420–3433, Dec. 2018.
- [110] L. Tang and Y. Yang, "Analysis of synchronized charge extraction for piezoelectric energy harvesting," *Smart Mater. Struct.*, vol. 20, no. 8, Jul. 2011, Art. no. 085022, doi: 10.1088/0964-1726/20/8/085022.
- [111] G. A. Rincón-Mora and S. Yang, "Tiny piezoelectric harvesters: Principles, constraints, and power conversion," *IEEE Trans. Circuits Syst.*, *Regul. Papers*, vol. 63, no. 5, pp. 639–649, May 2016.
- [112] S. Kundu, M. Liu, S.-J. Wen, R. Wong, and C. H. Kim, "A fully integrated digital LDO with built-in adaptive sampling and active voltage positioning using a beat-frequency quantizer," *IEEE J. Solid-State Circuits*, vol. 54, no. 1, pp. 109–120, Jan. 2019.

- [113] M. El-Nozahi, A. Amer, J. Torres, K. Entesari, and E. Sanchez-Sinencio, "High PSR low drop-out regulator with feed-forward ripple cancellation technique," *IEEE J. Solid-State Circuits*, vol. 45, no. 3, pp. 565–577, Mar. 2010.
- [114] J. Guo and K. N. Leung, "A 25 mA CMOS LDO with -85 dB PSRR at 2.5 MHz," in *Proc. IEEE Asian Solid-State Circuits Conf.*, Nov. 2013, pp. 381-384, doi: 10.1109/ASSCC.2013.6691062.
- [115] C.-J. Park, M. Onabajo, and J. Silva-Martinez, "External capacitor-less low drop-out regulator with 25 dB superior power supply rejection in the 0.4–4 MHz range," *IEEE J. Solid-State Circuits*, vol. 49, no. 2, pp. 486–501, Feb. 2014.
- [116] B. Yang, B. Drost, S. Rao, and P. K. Hanumolu, "A high-PSR LDO using a feedforward supply-noise cancellation technique," in *Proc. IEEE Custom Integr. Circuits Conf.*, Sep. 2011, pp. 1–4, doi: 10.1109/CICC.2011.6055409.
- [117] R. J. Milliken, J. Silva-Martinez, and E. Sanchez-Sinencio, "Full on-chip CMOS low-dropout voltage regulator," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 54, no. 9, pp. 1879–1890, Sep. 2007.
- [118] M. Cheah, D. Mandal, B. Bakkaloglu, and S. Kiaei, "A 100-mA, 99.11% current efficiency, 2-mVpp ripple digitally controlled LDO with active ripple suppression," *IEEE Trans. Very Large Scale Integr. VLSI Syst.*, vol. 25, no. 2, pp. 696–704, Feb. 2017.
- [119] Y.-F. Chen and K.-T. Tang, "A wireless power transmission subsystem with capacitor-less high PSR LDO and thermal protection mechanism for artificial retina application," in *Proc. VLSI Des., Autom. Test*, Apr. 2015, pp. 1–4, doi: 10.1109/VLSI-DAT.2015.7114516.
- [120] Y. Hu, B. Ma, H. Hao, and L. Li, "Intermediate multimedia node: Implantable spinal cord stimulator," J. Vis. Commun. Image Represent., vol. 41, pp. 15–20, Nov. 2016, doi: 10.1016/j.jvcir.2016.09.001.
- [121] S. Fan, Z. Xue, H. Lu, Y. Song, H. Li, and L. Geng, "Area-efficient on-chip DC–DC converter with multiple-output for bio-medical applications," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 61, no. 11, pp. 3298–3308, Nov. 2014.
- [122] H. Mo and D. Kim, "Multiple-output LDO regulator applying with constant feedback factor," in *Proc. Int. SoC Des. Conf.*, Nov. 2017, pp. 194–195, doi: 10.1109/ISOCC.2017.8368848.
- [123] M. A. Akram, W. Hong, and I.-C. Hwang, "Capacitorless self-clocked all-digital low-dropout regulator," *IEEE J. Solid-State Circuits*, vol. 54, no. 1, pp. 266–276, Jan. 2019.
- [124] S. Kaedi, M. B. Ghaznavi-Ghoushchi, and M. Rahimi, "A power efficient multi-level output all digital LDO with fast settling time and built in self calibration for DVFS and multi-VDD applications," in *Proc. 23rd Iranian Conf. Elect. Eng.*, May 2015, pp. 1276–1281, doi: 10.1109/IranianCEE.2015.7146412.
- [125] C.-Y. Wu, J.-C. Lou, and Z.-B. Deng, "An ultra-low power capacitorless LDO for always-on domain in NB-IoT applications," in *Proc. IEEE Int. Conf. Appl. Syst. Invention*, Apr. 2018, pp. 137–140, doi: 10.1109/ICASI.2018.8394550.
- [126] P. Hazucha, T. Karnik, B. A. Bloechel, C. Parsons, D. Finan, and S. Borkar, "Area-efficient linear regulator with ultra-fast load regulation," *IEEE J. Solid-State Circuits*, vol. 40, no. 4, pp. 933–940, Apr. 2005.
- [127] V. Nair and J. Choi, "An integrated chip high-voltage power receiver for wireless biomedical implants," *Energies*, vol. 8, pp. 5467–5487, Jun. 2015, doi: 10.3390/en8065467.
- [128] H. Jiang, Y. Guo, Z. Wu, C. Zhang, W. Jia, and Z. Wang, "Implantable wireless intracranial pressure monitoring based on air pressure sensing," *IEEE Trans. Biomed. Circuits Syst.*, vol. 12, no. 5, pp. 1076–1087, Oct. 2018.
- [129] S. R. Platt, S. Farritor, and H. Haider, "On low-frequency electric power generation with PZT ceramics," *IEEE/ASME Trans. Mechatron.*, vol. 10, no. 2, pp. 240–252, Apr. 2005.
- [130] P. Weng, H. Tang, P. Ku, and L. Lu, "50 mV-input batteryless boost converter for thermal energy harvesting," *IEEE J. Solid-State Circuits*, vol. 48, no. 4, pp. 1031–1041, Apr. 2013.
- [131] E. J. Carlson, K. Strunz, and B. P. Otis, "A 20 mV input boost converter with efficient digital control for thermoelectric energy harvesting," *IEEE J. Solid-State Circuits*, vol. 45, no. 4, pp. 741–750, Apr. 2010.
- [132] J. Im, S. Wang, S. Ryu, and G. Cho, "A 40 mV transformer-reuse selfstartup boost converter with MPPT control for thermoelectric energy harvesting," *IEEE J. Solid-State Circuits*, vol. 47, no. 12, pp. 3055–3067, Dec. 2012.
- [133] A. Shrivastava, N. E. Roberts, O. U. Khan, D. D. Wentzloff, and B. H. Calhoun, "A 10 mV-input boost converter with inductor peak current control and zero detection for thermoelectric and solar energy harvesting with 220 mV cold-start and -14.5 dBm, 915 MHz RF kick-start," *IEEE J. Solid-State Circuits*, vol. 50, no. 8, pp. 1820–1832, Aug. 2015.

- [134] D. Rozgić and D. Marković, "A miniaturized 0.78-mW/cm² autonomous thermoelectric energy-harvesting platform for biomedical sensors," *IEEE Trans. Biomed. Circuits Syst.*, vol. 11, no. 4, pp. 773–783, Aug. 2017.
- [135] J. Goeppert and Y. Manoli, "Fully integrated startup at 70 mV of boost converters for thermoelectric energy harvesting," *IEEE J. Solid-State Circuits*, vol. 51, no. 7, pp. 1716–1726, Jul. 2016.
- [136] Y. Zhang *et al.*, "A batteryless 19 μW MICS/ISM-Band energy harvesting body sensor node SoC for ExG applications," *IEEE J. Solid-State Circuits*, vol. 48, no. 1, pp. 199–213, Jan. 2013.
- [137] P. Chen and P. M. Fan, "An 83.4% peak efficiency single-inductor multiple-output based adaptive gate biasing DC-DC converter for thermoelectric energy harvesting," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 62, no. 2, pp. 405–412, Feb. 2015.
- [138] J. Katic, S. Rodriguez, and A. Rusu, "A dual-output thermoelectric energy harvesting interface with 86.6% peak efficiency at 30 μW and total control power of 160 nW," *IEEE J. Solid-State Circuits*, vol. 51, no. 8, pp. 1928–1937, Aug. 2016.
- [139] Y. K. Ramadass and A. P. Chandrakasan, "A battery-less thermoelectric energy harvesting interface circuit with 35 mV startup voltage," *IEEE J. Solid-State Circuits*, vol. 46, no. 1, pp. 333–341, Jan. 2011.
- [140] Y. Teh and P. K. T. Mok, "Design of transformer-based boost converter for high internal resistance energy harvesting sources with 21 mV self-startup voltage and 74% power efficiency," *IEEE J. Solid-State Circuits*, vol. 49, no. 11, pp. 2694–2704, Nov. 2014.
- [141] S. Carreon-Bautista, C. Erbay, A. Han, and E. Sánchez-Sinencio, "Power management system with integrated maximum power extraction algorithm for microbial fuel cells," *IEEE Trans. Energy Convers.*, vol. 30, no. 1, pp. 262–272, Mar. 2015.
- [142] X. Zhang, H. Ren, S. Pyo, J. Lee, J. Kim, and J. Chae, "A high-efficiency DC–DC boost converter for a miniaturized microbial fuel cell," *IEEE Trans. Power Electron.*, vol. 30, no. 4, pp. 2041–2049, Apr. 2015.
- [143] D. El-Damak and A. P. Chandrakasan, "A 10 nW–1 μW power management IC with integrated battery management and self-startup for energy harvesting applications," *IEEE J. Solid-State Circuits*, vol. 51, no. 4, pp. 943–954, Apr. 2016.
- [144] J. Katic, S. Rodriguez, and A. Rusu, "A high-efficiency energy harvesting interface for implanted biofuel cell and thermal harvesters," *IEEE Trans. Power Electron.*, vol. 33, no. 5, pp. 4125–4134, May 2018.
- [145] S. Bandyopadhyay and A. P. Chandrakasan, "Platform architecture for solar, thermal, and vibration energy combining with MPPT and single inductor," *IEEE J. Solid-State Circuits*, vol. 47, no. 9, pp. 2199–2215, Sep. 2012.
- [146] G. Chowdary, A. Singh, and S. Chatterjee, "An 18 nA, 87% efficient solar, vibration and RF energy-harvesting power management system with a single shared inductor," *IEEE J. Solid-State Circuits*, vol. 51, no. 10, pp. 2501–2513, Oct. 2016.
- [147] D. Ma, W.-H. Ki, C.-Y. Tsui, and P. K. T. Mok, "Single-inductor multiple-output switching converters with time-multiplexing control in discontinuous conduction mode," *IEEE J. Solid-State Circuits*, vol. 38, no. 1, pp. 89–100, Jan. 2003.
- [148] Y. H. Lam, W. H. Ki, and C. Y. Tsui, "Single inductor multipleinput multiple-output switching converter and method of use," U.S.7256568B2, Aug. 14, 2007.
- [149] F. U. Ahmed, Z. T. Sandhie, L. Ali, and M. H. Chowdhury, "A brief overview of on-chip voltage regulation in high-performance and high-density integrated circuits," *IEEE Access*, vol. 9, pp. 813–826, 2021.
- [150] M. D. Seeman, "A design methodology for switched-capacitor DC-DC converters," Defense Tech. Inf. Center, Fort Belvoir, VA, USA, May 2009, doi: 10.21236/ADA538398.
- [151] L. George, G. D. Gargiulo, T. Lehmann, and T. J. Hamilton, "A 0.04 mm² buck-boost DC-DC converter for biomedical implants using adaptive gain and discrete frequency scaling control," *IEEE Trans. Biomed. Circuits Syst.*, vol. 10, no. 3, pp. 668–678, Jun. 2016.
- [152] O. A.-T. Hasib, M. Sawan, and Y. Savaria, "A low-power asynchronous step-down DC–DC converter for implantable devices," *IEEE Trans. Biomed. Circuits Syst.*, vol. 5, no. 3, pp. 292–301, Jun. 2011.
- [153] P. Nadeau and M. Sawan, "A flexible high voltage biphasic currentcontrolled stimulator," in *Proc. IEEE Biomed. Circuits Syst. Conf.*, Nov. 2006, pp. 206–209.
- [154] M. Wieckowski, G. K. Chen, M. Seok, D. Blaauw, and D. Sylvester, "A hybrid DC-DC converter for sub-microwatt sub-1V implantable applications," in *Proc. Symp. VLSI Circuits*, Jun. 2009, pp. 166–167.
- [155] J. F. Dickson, "On-Chip high-voltage generation in integrated circuits using an improved multiplier technique," *IEEE J. Solid-State Circuits*, vol. 11, no. 3, pp. 374–378, Jun. 1976.

- [156] Z. Chen, M. Law, P. Mak, and R. P. Martins, "A single-chip solar energy harvesting IC using integrated photodiodes for biomedical implant applications," *IEEE Trans. Biomed. Circuits Syst.*, vol. 11, no. 1, pp. 44–53, Feb. 2017.
- [157] R. Pelliconi, D. Iezzi, A. Baroni, M. Pasotti, and P. L. Rolandi, "Power efficient charge pump in deep submicron standard CMOS technology," *IEEE J. Solid-State Circuits*, vol. 38, no. 6, pp. 1068–1071, Jun. 2003.
- [158] M. S. Makowski, "Realizability conditions and bounds on synthesis of switched-capacitor DC-DC voltage multiplier circuits," *IEEE Trans. Circuits Syst. Fundam. Theory Appl.*, vol. 44, no. 8, pp. 684–691, Aug. 1997.
- [159] D. Somasekhar *et al.*, "Multi-phase 1GHz voltage doubler charge-pump in 32nm logic process," in *Proc. Symp. VLSI Circuits*, Jun. 2009, pp. 196–197.
- [160] Z. Safarian and H. Hashemi, "Capacitance-sharing, dual-output, compact, switched-capacitor DC–DC converter for low-power biomedical implants," *Electron. Lett.*, vol. 50, no. 23, pp. 1673–1675, 2014, doi: 10.1049/el.2014.1842.
- [161] H. Chen *et al.*, "Low-power circuits for the bidirectional wireless monitoring system of the orthopedic implants," *IEEE Trans. Biomed. Circuits Syst.*, vol. 3, no. 6, pp. 437–443, Dec. 2009.
- [162] D. Kilani, B. Mohammad, M. Alhawari, H. Saleh, and M. Ismail, "A dual-output switched capacitor DC–DC buck converter using adaptive time multiplexing technique in 65-nm CMOS," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 65, no. 11, pp. 4007–4016, Nov. 2018.
- [163] Z. Safarian and H. Hashemi, "Event-driven implantable neural recording integrated system using level-crossing detectors," in *Proc. IEEE Biomed. Circuits Syst. Conf.*, Oct. 2015, pp. 1–4, doi: 10.1109/Bio-CAS.2015.7348352.
- [164] I. Chowdhury and D. Ma, "Design of reconfigurable and robust integrated SC power converter for self-powered energy-efficient devices," *IEEE Trans. Ind. Electron.*, vol. 56, no. 10, pp. 4018–4028, Oct. 2009.
- [165] C. Zheng, L. Su, and D. Ma, "A systematic USFG design approach for integrated reconfigurable switched-capacitor power converters," *IEEE Trans. Circuits Syst., Regul. Papers*, vol. 58, no. 11, pp. 2790–2800, Nov. 2011.
- [166] A. Rao, W. McIntyre, U.-K. Moon, and G. C. Temes, "Noise-shaping techniques applied to switched-capacitor voltage regulators," *IEEE J. Solid-State Circuits*, vol. 40, no. 2, pp. 422–429, Feb. 2005.
- [167] G. V. Piqué, "A 41-phase switched-capacitor power converter with 3.8 mV output ripple and 81% efficiency in baseline 90 nm CMOS," in *Proc. IEEE Int. Solid-State Circuits Conf.*, Feb. 2012, pp. 98–100, doi: 10.1109/ISSCC.2012.6176892.
- [168] K. N. Leung and P. K. T. Mok, "A capacitor-free CMOS low-dropout regulator with damping-factor-control frequency compensation," *IEEE J. Solid-State Circuits*, vol. 38, no. 10, pp. 1691–1702, Oct. 2003.
- [169] Y. K. Ramadass, A. A. Fayed, and A. P. Chandrakasan, "A fully-integrated switched-capacitor step-down DC-DC converter with digital capacitance modulation in 45 nm CMOS," *IEEE J. Solid-State Circuits*, vol. 45, no. 12, pp. 2557–2565, Dec. 2010.
- [170] H. Le, S. R. Sanders, and E. Alon, "Design techniques for fully integrated switched-capacitor DC-DC converters," *IEEE J. Solid-State Circuits*, vol. 46, no. 9, pp. 2120–2131, Sep. 2011.
- [171] J. H. Klootwijk *et al.*, "Ultrahigh capacitance density for multiple ALD-grown MIM capacitor stacks in 3-D silicon," *IEEE Electron Device Lett.*, vol. 29, no. 7, pp. 740–742, Jul. 2008.
 [172] Z. Hayashi, Y. Katayama, M. Edo, and H. Nishio, "High-efficiency dc-
- [172] Z. Hayashi, Y. Katayama, M. Edo, and H. Nishio, "High-efficiency dcdc converter chip size module with integrated soft ferrite," *IEEE Trans. Magn.*, vol. 39, no. 5, pp. 3068–3072, Sep. 2003.
- [173] G. Schrom et al., "A 100 MHz eight-phase buck converter delivering 12 A in 25 mm² using air-core inductors," in Proc. 22nd Annu. IEEE Appl. Power Electron. Conf. Expo., Feb. 2007, pp. 727–730, doi: 10.1109/APEX.2007.357595.
- [174] K. Onizuka, K. Inagaki, H. Kawaguchi, M. Takamiya, and T. Sakurai, "Stacked-chip implementation of on-chip buck converter for distributed power supply system in SiPs," *IEEE J. Solid-State Circuits*, vol. 42, no. 11, pp. 2404–2410, Nov. 2007.
- [175] M. Wens and M. Steyaert, "A fully-integrated 0.18 μm CMOS DC-DC step-down converter, using a bondwire spiral inductor," in *Proc. IEEE Custom Integr. Circuits Conf.*, Sep. 2008, pp. 17–20, doi: 10.1109/CICC.2008.4672009.
- [176] H. J. Bergveld, R. Karadi, and K. Nowak, "An inductive down converter system-in-package for integrated power management in battery-powered applications," in *Proc. IEEE Power Electron. Spec. Conf.*, Jun. 2008, pp. 3335–3341, doi: 10.1109/PESC.2008.4592470.

- [177] D. Maksimovic and S. Dhar, "Switched-capacitor DC-DC converters for low-power on-chip applications," in *Proc. 30th Annu. IEEE Power Electron. Spec. Conf. Rec.*, Jul. 1999, vol. 1, pp. 54–59, doi: 10.1109/PESC.1999.788980.
- [178] J. Kim, P. K. T. Mok, and C. Kim, "A 0.15 v input energy harvesting charge pump with dynamic body biasing and adaptive dead-time for efficiency improvement," *IEEE J. Solid-State Circuits*, vol. 50, no. 2, pp. 414–425, Feb. 2015.
- [179] W. Jung *et al.*, "An ultra-low power fully integrated energy harvester based on self-oscillating switched-capacitor voltage doubler," *IEEE J. Solid-State Circuits*, vol. 49, no. 12, pp. 2800–2811, Dec. 2014.
- [180] Y.-C. Shih and B. P. Otis, "An inductorless DC–DC converter for energy harvesting with a 1.2- μW bandgap-referenced output controller," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 58, no. 12, pp. 832–836, Dec. 2011.
- [181] I. Doms, P. Merken, C. Van Hoof, and R. P. Mertens, "Capacitive power management circuit for micropower thermoelectric generators with a 1.4 μA controller," *IEEE J. Solid-State Circuits*, vol. 44, no. 10, pp. 2824–2833, Oct. 2009.
- [182] T. Ozaki, T. Hirose, T. Nagai, K. Tsubaki, N. Kuroki, and M. Numa, "A 0.21-V minimum input, 73.6% maximum efficiency, fully integrated 3-terminal voltage converter with MPPT for low-voltage energy harvesters," in *Proc. 20th Asia South Pacific Des. Automat. Conf.*, Jan. 2015, pp. 30–31.
- [183] J. Kim, J. Kim, and C. Kim, "A regulated charge pump with a lowpower integrated optimum power point tracking algorithm for indoor solar energy harvesting," *IEEE Trans. Circuits Syst. II, Express Briefs*, vol. 58, no. 12, pp. 802–806, Dec. 2011.
- [184] T. Ozaki, T. Hirose, H. Asano, N. Kuroki, and M. Numa, "Fully-integrated high-conversion-ratio dual-output voltage boost converter with MPPT for low-voltage energy harvesting," *IEEE J. Solid-State Circuits*, vol. 51, no. 10, pp. 2398–2407, Oct. 2016.
- [185] N. Sturcken *et al.*, "A 2.5D integrated voltage regulator using coupledmagnetic-core inductors on silicon interposer delivering 10.8 A/mm²," in *Proc. IEEE Int. Solid-State Circuits Conf.*, Feb. 2012, pp. 400–402, doi: 10.1109/ISSCC.2012.6177064.
- [186] Y. Qiu, C. Van Liempd, B. O. Het Veld, P. G. Blanken, and C. Van Hoof, "5 μW-to-10 mW input power range inductive boost converter for indoor photovoltaic energy harvesting with integrated maximum power point tracking algorithm," in *Proc. IEEE Int. Solid-State Circuits Conf.*, Feb. 2011, pp. 118–120.
- [187] M. Dini, A. Romani, M. Filippi, and M. Tartagni, "A nanocurrent power management IC for low-voltage energy harvesting sources," *IEEE Trans. Power Electron.*, vol. 31, no. 6, pp. 4292–4304, Jun. 2016.
- [188] P.-H. Chen et al., "An 80 mV startup dual-mode boost converter by charge-pumped pulse generator and threshold voltage tuned oscillator with hot carrier injection," *IEEE J. Solid-State Circuits*, vol. 47, no. 11, pp. 2554–2562, Nov. 2012.
- [189] P.-H. Chen *et al.*, "Startup techniques for 95 mV step-up converter by capacitor pass-on scheme and V_{th}-tuned oscillator with fixed charge programming," *IEEE J. Solid-State Circuits*, vol. 47, no. 5, pp. 1252–1260, May 2012.
- [190] N. Sze, W. Ki, and C. Tsui, "Threshold voltage Start-up boost converter for sub-mA applications," in *Proc. 4th IEEE Int. Symp. Electron. Des.*, *Test Appl.*, Jan. 2008, pp. 338–341, doi: 10.1109/DELTA.2008.126.
- [191] P. Garcha et al., "A 25 mV-startup cold start system with on-chip magnetics for thermal energy harvesting," in *Proc. 43rd IEEE Eur. Solid State Circuits Conf.*, Sep. 2017, pp. 127–130, doi: 10.1109/ESS-CIRC.2017.8094542.
- [192] A. M. Imtiaz, F. H. Khan, and J. S. Walling, "Contour-mode ringshaped AlN microresonator on Si and feasibility of its application in series-resonant converter," *IEEE Trans. Power Electron.*, vol. 30, no. 8, pp. 4437–4454, Aug. 2015.
- [193] A. M. Imtiaz and F. H. Khan, "Film bulk acoustic resonator (FBAR) based power converters: A new trend featuring EMI reduction and high power density," in *Proc. 28th Annu. IEEE Appl. Power Electron. Conf. Expo.*, Long Beach, CA, USA, 2013, pp. 1485–1491, doi: 10.1109/APEC.2013.6520494.
- [194] A. N. M. Wasekul Azad, S. Roy, A. S. Imtiaz, and F. Khan, "Microelectromechanical system (MEMS) resonator: A new element in power converter circuits featuring reduced EMI," in *Proc. Int. Power Electron. Conf.*, May 2018, pp. 2416–2420.



Sourov Roy (Graduate Student Member, IEEE) received the B.Sc. degree in electrical and electronic engineering from the Bangladesh University of Engineering and Technology, Dhaka, Bangladesh, in 2012, and the M.S. and Ph.D. degrees from the University of Missouri-Kansas City, Kansas City, MO, USA, in 2020 and 2021, respectively, both in electrical engineering.

He is working as a Postdoctoral Associate with the Missouri Institute for Defense and Energy, University of Missouri—Kansas City. Prior to his cur-

rent endeavors, he was a Lecturer with Daffodil International University in Dhaka, Bangladesh, from 2012 to 2013. Then, from 2013 to 2014, he was with Civil Aviation Authority, Bangladesh, as an Assistant Engineer. He also served Bangladesh Power Development Board and Dhaka Power Distribution Company Limited, Bangladesh, as an Assistant Engineer, from 2014 to 2015. His research interests include energy harvesting and power conditioning circuits for biomedical applications, fault diagnosis and condition monitoring of energy systems and power electronics, and pulse-power and high-power microwave systems.



Somen Baidya (Student Member, IEEE) received the B.Sc. degree in electrical and electronic engineering from the Bangladesh University of Engineering and Technology, Dhaka, Bangladesh, in 2012, and the M.Sc. degree in applied engineering from the Georgia Southern University, Statesboro, GA, USA, in 2016. He is currently working toward the Ph.D. degree with the Department of Computer Science and Electrical Engineering, University of Missouri—Kansas City, Kansas City, MO, USA.

From 2012 to 2014, he was with Robi Axiata Limited, as a System Engineer. His current research interests include bioelectromagnetics, modeling of the electrical activities of cancerous cells, plasmonic biomarker, and optical scattering characterization of lunar regolith particles.

Mr. Baidya has been an active member of the Biomedical Engineering Society since 2018.



A N M Wasekul Azad (Graduate Student Member, IEEE) received the B.Sc. degree in electrical and electronic engineering from the Bangladesh University of Engineering and Technology, Dhaka, Bangladesh, in 2012. He is currently working toward the Ph.D. degree with the Department of Computer Science and Electrical Engineering, University of Missouri— Kansas City, Kansas City, MO, USA.

He served Robi Axiata, Ltd., as a Central Network Configuration Engineer in Bangladesh, from 2012 to 2016. His research interests include widebandgap

power semiconductor devices, high-voltage switch design, gate drive techniques, and pulsed power sources.



Faisal Khan (Senior Member, IEEE) received the B.Sc. degree from the Bangladesh University of Engineering and Technology, Dhaka, Bangladesh, in 1999, the M.S. degree from Arizona State University, Tempe, AZ, USA, in 2003, and the Ph.D. degree from the University of Tennessee, Knoxville, TN, USA, in 2007, all in electrical engineering.

From 2007 to 2009, he was with the Electric Power Research Institute as a Senior Power Electronics Engineer. From 2009 to 2014, he was with the Electrical and Computer Engineering Department, University

of Utah, as an Assistant Professor. Since 2015, he has been with the Department of Computer Science and Electrical Engineering, University of Missouri— Kansas City, Kansas City, MO, USA, as an Associate Professor. His major research interests include characterizing degradation/aging in power converters using spread-spectrum time-domain reflectometry, along with micropower converters for biomedical applications.

Prof. Khan is currently an Associate Editor for the IEEE TRANSACTIONS ON POWER ELECTRONICS.